Review of the Effectiveness and Cost Effectiveness of Interventions, Strategies, Programmes and Policies to Help Recipients of Incapacity Benefits Return to Employment (Paid and Unpaid)

> Report May 2008

Evidence review 3

Review of the Effectiveness and Cost Effectiveness of Interventions, Strategies, Programmes and Policies to Help Recipients of Incapacity Benefits Return to Employment (Paid and Unpaid)

Sue Hayday Dr Jo Rick Dr Christopher Carroll Nick Jagger and Jim Hillage



The School Of University Of And Sheffield. Related Research





Published by:

INSTITUTE FOR EMPLOYMENT STUDIES Mantell Building University of Sussex Falmer Brighton BN1 9RF UK

Tel. + 44 (0) 1273 686751 Fax + 44 (0) 1273 690430

http://www.employment-studies.co.uk

Copyright © 2008 Institute for Employment Studies

No part of this publication may be reproduced or used in any form by any means graphic, electronic or mechanical including photocopying, recording, taping or information storage or retrieval systems—without prior permission in writing from the Institute for Employment Studies.

The Institute for Employment Studies

The Institute for Employment Studies is an independent, apolitical, international centre of research and consultancy in public employment policy and organisational human resource issues. For 40 years the Institute has been a focus of knowledge and practical experience in employment and training policy, the operation of labour markets and human resource planning and development. IES is a not-for-profit organisation and has over 60 multidisciplinary staff and international associates. IES aims to help bring about sustainable improvements in employment policy and human resource management by increasing the understanding and improving the practice of key decision makers in policy bodies and employing organisations.

Institute of Work Psychology, Sheffield University

The Institute of Work Psychology is dedicated to conducting applied research in work settings, in both the public and private sectors. The aims of the Institute are to:

- advance knowledge about the causes of individual, team and organisational effectiveness at work
- increase understanding of the well-being of people at work
- advance knowledge about innovation and creativity at work
- disseminate this knowledge in the scientific community, in the workplace and in the wider public domain
- design, implement and evaluate methods of promoting effectiveness, innovation and well-being at work.

The School of Health and Related Research, Sheffield University

ScHARR is a large multi-disciplinary School within the Faculty of Medicine at the University of Sheffield. It employs around 200 staff, primarily behavioural and social scientists, epidemiologists, statisticians, public health specialists and health economists. Its research infrastructure and its multi-disciplinarily approach offer huge potential for collaboration with leading researchers in other fields, whilst its Information Resources Section is able to provide first class support with systematic literature searches. ScHARR staff have considerable expertise in systematic review methodology and have contributed to the development of the latest techniques for identifying, assessing and synthesising non RCT evidence in systematic review methodology.

Content

Exe	ecutiv	e Summary	ix
	Initial title and abstract sifting Full paper screening Included papers Cognitive behavioural therapy Rehabilitation programme Conclusions		xi xi xi xi xi xii xii
1.	Introduction		1
	1.1 1.2 1.3	Background Research objectives Structure of report	1 2 3
2	Methodology		
	 2.1 2.2 2.3 2.4 2.5 	Identifying potentially relevant studies Selection of studies for inclusion Summary of studies identified for inclusion Data extraction and quality appraisal Synthesis and formulation of evidence statements	4 7 16 20 23
3	Effectiveness Findings		
	3.1 3.2 3.3 3.4	Theme 1: Cognitive behavioural therapy Theme 2: Rehabilitation programme Evidence tables for the two themes Applicability of the evidence to the UK populations in the scope	24 27 33 39
4	Cost Effectiveness Findings		40
	4.1 4.2	Cost–effectiveness results Applicability of the evidence to the UK populations in the scope	40 41
5	Disc	russion	42

References	45
Appendix 1: Example Search Strategy Used for Research Question 4	46
Appendix 2: Sifting Criteria Used	48
Appendix 3: Full Paper Screening Checklists	56
Appendix 4: Excluded Primary Studies by Reason for Exclusion	62
Appendix 5: Q4 Primary Studies Pending	77
Appendix 6: Example Completed Effectiveness Data Extraction and Quality Checklists	78
Appendix 7: Economic Data Extraction and Quality Assessment Form	90

Executive Summary

The National Institute for Health and Clinical Excellence (NICE) has been asked by the Department of Health to develop guidance for primary care services and employers on the management of long-term sickness and incapacity. The guidance will provide recommendations for good practice that are based on the best available evidence of effectiveness and cost effectiveness.

This report is one in a series of reviews of the literature covering primary studies of interventions, strategies, programmes and policies to help those in receipt of incapacity benefit return to work. As such, the report aims to inform the guidance through two linked systematic reviews of the literature on the effectiveness and the cost effectiveness of interventions to move incapacity benefit recipients into employment.

Specifically, this review addressed the following primary research question:

'What UK primary care-based interventions, programmes, policies or strategies are effective and cost-effective in helping those in receipt of incapacity benefit to return to full or part time employment? These could be delivered by a number of sectors (such as voluntary or education sectors) in collaboration with, and/or funded by, employers and primary care services.'

A protocol was developed which specified the population, interventions and outcomes of interest for the effectiveness and cost effectiveness reviews. The protocol provided the detailed inclusion and exclusion criteria that were applied to the literature retrieved via the online searches specified in the protocol. In addition to these database and website searches, experts and the Programme Development Group were contacted for addition studies. The citations of and references given in the included papers were also checked. The process took into account the following included and excluded population groups.

Populations covered:

- All adults over the age of 16 living in the community who are unemployed because of long-term incapacity and/or in receipt of incapacity benefit/disability benefits or other similar benefit.
- Locations to be included:
 - UK only studies as incapacity benefit (and any previous benefits) designates a specific UK based policy
- Populations excluded were:
 - □ Unemployment benefit recipients
 - □ Employed/self employed
- The settings excluded were:
 - □ Studies conducted in non-UK settings
- The interventions, programmes, policies and strategies included were any that:
 - Could help adults in receipt of incapacity benefit or have helped recipients of earlier forms of the benefit to return to work (paid/unpaid) or prepare for work (paid/unpaid).
 - Interventions, policies, programmes or strategies delivered in a primary-care setting and/or workplace setting and/or planned, designed, delivered, managed or funded in collaboration with primary care providers and/or employers. These interventions, policies, programmes or strategies can be delivered by a number of providers (such as voluntary, private, statutory sectors) and/or in various settings not just workplace or primary care settings (such as job centres, community centres) as long as they are fully or co-planned, designed, delivered, managed and/or funded in collaboration with employers and primary care services.
- Interventions, programmes, policies and strategies excluded were any that:
 - Dealt solely with the provision of treatment for existing conditions (including pharmacological or therapeutic interventions)
 - Dealt solely with the effectiveness of the incapacity benefit system, private health insurance schemes or statutory or occupational sick pay.
 - Dealt solely with preventing ill-health retirement (ie where recipient has no intention of returning to work).

Based on the protocol, searches were undertaken of 19 research and specialist economics data bases and six websites by the Centre for Review and Dissemination (CRD) at York University.

Initial title and abstract sifting

A total of 5,899 articles were identified (5,546 effectiveness and 353 cost effectiveness), supplemented by website searches, relevant references from review articles, suggestions from experts who have been consulted about the available literature, citation searches and checking references from the final included papers.

The titles and abstracts of these articles were all initially sifted against the agreed inclusion and exclusion criteria from the protocol. Papers definitely meeting the criteria were put forward for full paper screening. Those where it was unclear if a study met all the criteria were tagged as 'get full paper' and put forward for full paper screening. Those relevant to this review from the searches for the forthcoming reviews were tagged and screened against appropriate sift criteria labelled as 'include' or 'get full paper'. Otherwise papers were excluded from the review. A total of 180 primary effectiveness and cost effectiveness papers from all sources were ordered and retrieved for full paper screening.

Full paper screening

The full paper screening involved a more thorough check of the studies suitability for inclusion in the review. This screening was undertaken using full paper screening checklist based on the agreed protocol. Given that the decisions were based on the full papers rather than simply the title and abstract, and in some cases only the title, more definitive decisions could be made. In addition all the papers were subject to a second review at the full paper screening stage to validate the decisions made.

Included papers

Articles passing the full paper screen were then put though a process of data extraction and quality assessment. Data extraction was performed by one reviewer and checked by another. Quality assessment was undertaken by two reviewers independently and ratings of quality were then compared. Any differences were settled through discussion. Three effectiveness articles and no cost-effectiveness articles passed this full paper screen. The papers covered different issues were reviewed under three themes.

Intervention Theme 1: Cognitive behavioural therapy

Evidence statement for Theme 1

ER3.1: There is insufficient evidence from one RCT (grade '-') to assess the efficacy of cognitive behavioural therapy delivered by Jobcentre Plus work psychologists over an eight week period in

Intervention Theme 2: Work-focussed interviews and access to employability support

Evidence statement for Theme 2

ER3.2: There is limited evidence from a non-randomised controlled trial (grade +) that a programme comprising attendance at a work-focussed interview and access to return to work support (including further interviews, help with managing their health condition, financial support and in-work occupational health and personal support) could be effective at increasing the chances of people on Incapacity Benefit (IB) being in work 18 months after initially enquiring about accessing IB. The employment effects appear to be stronger for women than men, those aged under, rather than over, 50 and people without rather than with mental illness. (Bewley et al. 2007)

Intervention Theme 3: Rehabilitation programme

Evidence statement for Theme 3

ER3.3: There is insufficient evidence from one case series study (grade -) to assess the efficacy of the Papworth Trust's Early Rehabilitation Programme (comprising support for participants over a four to ten month period from a rehabilitation coordinator, case manager, job coach and assistant, vocational adviser, information technology assessor and vocational psychologist - help from a literacy tutor, an occupational therapist and assistant, a consultant in rehabilitative medicine, a speech and language therapist and a physiotherapist are also used according to individual need) in assisting the return to work of those on incapacity benefit because of disabling injuries (Desouza et al 2007).

Conclusions

The search for the literature was comprehensive and considerable effort was made to limit bias in the identification, selection, extraction and appraisal of the literature. Overall, this review has identified a paucity of evidence in relation to the effectiveness of interventions and a complete lack of evidence for the cost effectiveness of interventions that met the requirements of the research question and the inclusion and exclusion criteria (eg UK studies based on an RCT or longitudinal methodology).

Two conclusions are apparent. Firstly that the effectiveness and cost effectiveness reviews for research question 4 have identified a need to generate new research in this

area as there is so little research which meets the specified inclusion criteria and evaluates the impact of interventions, strategies, programmes and policies to aid those on incapacity benefits to return to work. In particular it would be useful to have some form of economic evaluation of interventions that appear to be effective (in this case the only example being the Pathways to Work pilot reported on by Bewley et al (2007)).¹

This clearly represents a gap in the evidence base, at least in terms of the types of primary studies included in this review. It can be concluded that there is insufficient UK information of this sort on which to base detailed policy and practice, but that this report identifies clear gaps for future research.

In the absence of a sufficient research base, one option is to take into account evidence from other OECD countries. The different contexts and benefit regimes operating in other countries will undoubtedly limit the transferability of any findings (and this was one of the reasons that this review focussed just on the UK). However, given the time it would take to build up a substantial body of evidence, future reviews of this area may benefit from developing a way of at least drawing out the principles operating in other countries to see if any lessons can be learnt for the UK

¹ Reference to a cost benefit analysis is made in the range of literature about Pathways to Work, but no published study was found during searching

1. Introduction

The National Institute for Health and Clinical Excellence (NICE) has been asked by the Department of Health to develop guidance for primary care services and employers on the management of long-term sickness and incapacity. The guidance will provide recommendations for good practice that are based on the best available evidence of effectiveness and cost effectiveness. The Institute for Employment Studies, Institute of Work Psychology and the School of Health and Related Research (both at Sheffield University) were contracted to undertake a series of three effectiveness and cost effectiveness reviews of primary randomised controlled or longitudinal studies (covering four research questions) and an economic analysis of the evidence to support the production of this guidance.

This report covers the effectiveness and cost-effectiveness review for research question 4 about what works to help those in receipt of Incapacity Benefit to return to full or part time employment, first discussed at the Programme Development Group (PDG) in December 2007. Separate reviews covered the other three primary research questions about various aspects of long-term sickness absence and were first considered by the PDG in February 2008 and April 2008.

This review has been revised in the light of the comments received from the PDG and any further evidence received through the search and sifting process since the first draft was completed.

An economic analysis, including economic modelling, will be presented at the PDG meeting in May 2008. This will cover a selection of topics identified in the reviews which have been chosen by the PDG and where there is sufficient data to make modelling feasible.

1.1 Background

Extensive studies of work and unemployment support the concept that work is beneficial for health and well-being. There is broad consensus across multiple

disciplines, disability groups, employers, unions, insurers and all political parties, based on extensive clinical evidence that, when their health conditions permit, sick and disabled people should be encouraged to remain in or to (re)-enter work as soon as possible. The benefits of work for these groups are that it is therapeutic, leads to better health outcomes and minimises the harmful effects of long-term sickness absence and the risk of long-term incapacity (Waddell and Burton, 2006). Prolonged absence can result in job loss and the longer a person is on incapacity benefit, the less likely they are to return to work (Cabinet Office, 2004). It has been estimated that for a person claiming incapacity benefit for a year, the average duration of their claim will be for eight years and after two years they are more likely to die or retire than return to work (DWP/DH/HSE, 2005). It is against this background the Department of Health is seeking to provide guidance on which interventions are effective and cost-effective in helping those in receipt of incapacity benefit to return to work. The guidance is intended to be used by professionals and managers who have public health as part of their remit working in the NHS, local authorities and the wider public, private, voluntary and community sectors.

1.2 Research objectives

This review addresses the following specific research question which is referred to as 'research question 4' throughout the report:

'What UK primary care-based interventions, programmes, policies or strategies are effective and cost-effective in helping those in receipt of incapacity benefit to return to full or part time employment? These could be delivered by a number of sectors (such as voluntary or education sectors) in collaboration with, and/or funded by, employers and primary care services.'

The following secondary research questions were developed to interrogate the data further (data permitting):

- What is the frequency, content, length and duration of an effective intervention, programme, policy or strategy?
- Which are the most effective, cost effective and acceptable interventions, programmes, policies or strategies for different groups? (eg age, conditions, gender, ethnic groups or social classes)
- Does the effectiveness of an intervention, programme, policy or strategy depend on the person leading it? (What are the significant characteristics of an effective leader: what training and skills are required?)
- What are the barriers to-and facilitators of-effective implementation?
- Does the intervention, programme, policy or strategy lead to any adverse or unintended (positive and negative) outcomes?

Which interventions, programmes, policies or strategies are ineffective and/or are not cost effective?

It is important to recognise that any evidence subsequently presented in relation to the secondary research questions is drawn from a limited pool of studies and cannot be considered on the same level as evidence about the primary outcome.

1.3 Structure of report

The structure of this report is as follows:

- Chapter 2 discusses how the literature search was conducted, the retrieval of papers, the selection of studies for inclusion, data extraction and quality assessment.
- Chapter 3 presents the effectiveness findings by theme/area
- Chapter 4 provides the cost-effectiveness findings
- Chapter 5 discusses the review findings, highlighting their applicability, limitations and any gaps.

Seven appendices present supporting documents.

2 Methodology

This chapter details the methodology for identifying studies for inclusion in review question four. First the search strategies for the effectiveness and cost effectiveness questions are given, with details of the data bases and websites searched. The methods for title and abstract screening are described along with inclusion and exclusion criteria used. The process for contacting experts and members of the PDG is explained. Next the full paper screening process is detailed, including the review level papers, and the additional inclusion and exclusion criteria used at this stage are given. Finally the data extraction and quality assessment is presented and a summary of included papers given.

2.1 Identifying potentially relevant studies

In consultation with NICE it was decided to conduct three strands of searches for research question 4¹:

- 1. effectiveness primary studies
- 2. effectiveness reviews; and
- 3. cost effectiveness primary studies and reviews.

Nineteen databases and six websites were searched by the Centre for Reviews and Dissemination (CRD) at York University, using a search protocol supplied by the IES/University of Sheffield team.

An additional five website searches were undertaken by NICE following suggestions received from the PDG.

¹ The search strategy for the remaining research questions (1 to 3) are covered in the other two evidence reviews.

2.1.1 Effectiveness literature searches

The following key terms and limiters were used in the search. Key search terms were defined as 'benefit recipients', 'return to work', 'work readiness' or other synonyms. The search strategies for the Medline search can be seen in Appendix 1.

The following limits were placed on the search strategy:

- Published from 1990 onwards
- Published in English language only.

The following 19 databases and six websites were searched to identify primary studies and review-level studies for this effectiveness rapid review:

Databases

- Cochrane Database of Systematic Reviews (CDSR)
- Database of Abstracts of Reviews of Effectiveness (DARE)
- Cochrane Central Register of Controlled Trials
- MEDLINE
- EMBASE
- PsycINFO
- CINAHL (Cumulative Index of Nursing and Allied Health Literature)
- AMED (Allied and Complementary Medicine)
- Business Source Premier
- British Nursing Index
- NHS HTA
- ASSIA (Applied Social Science Index and Abstracts)
- Social Science Citation Index
- Science Citation Index
- Sociological Abstracts

Four databases of grey literature were also searched:

 HMIC (Health Management Information Consortium. Comprises King's Fund Database and DH-Data database)

- SIGLE (International System for grey literature)
- National Research Register
- Current Contents.

Websites

A series of websites were also searched to identify any relevant literature:

- Department for Work & Pensions: http://www.dwp.gov.uk/
- Institute for Public Policy & Research: http://www.ippr.org.uk/
- Employment studies research unit: http://www.uwe.ac.uk/bbs/research/esru/wps.shtml
- Centre for Longitudinal Studies: http://www.cls.ioe.ac.uk/
- Health and Safety Executive: http://www.hse.gov.uk/index.htm
- Department for Business, Enterprise and Regulatory Reform: http://www.dti.gov.uk/index.html

The database and website effectiveness search for research question 4 resulted in 5,627 primary study titles and abstracts¹ and a further 716 review-level references were also identified.

2.1.2 Cost Effectiveness literature searches

Specific economic searches, with the same limitations as for the effectiveness searches, were performed on following specialist economic databases:

- Health Economics Evaluation Database (HEED)
- NHS EED (NHS Economics Evaluation Database)
- Econlit.

The cost effectiveness search identified 353 titles and abstracts (all primary studies) for research question 4.

¹ That is 5,546 from the database searches and 81 from the website searches.

2.1.3 Suggestions from experts and PDG

The IES/University of Sheffield team put together a list of experts in the area of managing sickness absence and return to work. The list included academics working in the field, policy makers and commentators. This list, together with the titles of articles identified in the question four searches and the website searches that had passed the first stage of screening (see 2.2.1. for further details), were circulated to PDG members by NICE in October 2007. PDG members were asked to suggest additional experts who should be contacted and for any references they felt relevant to the review, which were not on the list.

Experts suggested by IES/University of Sheffield, NICE and the PDG were then all contacted with the titles of articles identified as relevant (following the title and abstract screening) to research question 4 and asked to suggest any additional references.

This resulted in an additional (to the above database and website searches) 21 effectiveness titles and abstracts. No economic titles and abstracts for research question 4 were suggested additional to those already identified.

2.1.4 Additional web-site searches

Following consultation with the PDG a further five website searches were undertaken:

- ACAS
- Institute of Occupational Health
- Oxford Health Alliance
- National Audit Office
- Xpert HR

The above website searches resulted in a further 13 effectiveness titles and abstracts and no additional economic titles/abstracts for research question 4 being identified (see section 2.2.4 below).

2.2 Selection of studies for inclusion

2.2.1 Title and abstract appraisal

The purpose of title and abstract appraisal is to identify studies that 'help to answer the questions being addressed by the review' (NHS CRD, 2001). Study selection should be systematic, replicable and free from bias. Sifting is informed by inclusion criteria reflecting the population, intervention, outcome and study design, and by exclusion criteria defined in the scoping document, such as language and date.

Only the criteria relevant to research question 4 are presented here. The criteria for the remaining research questions (1-3) are presented in the two other evidence reviews.

The inclusion criteria are set are set out below.

Settings included covered:

 UK only because incapacity benefit designates a specific population that only exists in the UK

Population included covered:

- all adults over the age of 16 who are:
 - unemployed because of long-term incapacity and in receipt of incapacity benefit/disability benefits.

Interventions, programmes, policies and strategies included covered:

 any that aimed to help adults in receipt of incapacity benefit to return to work or prepare for work.

Outcomes included covered:

- return to work after being in receipt of incapacity benefit
- job related activity (eg job seeking)
- other work related outcome.

Study designs that were included:

- Primary level study designs:
 - □ randomised controlled trials (RCTs)
 - longitudinal intervention studies (ie there is at least one follow up measure after baseline).
- Review level studies:
 - □ reviews of RCT's or longitudinal studies .

The sifting was limited to these study designs in line with the requirements of the research question. The aim of the sifting was to identify studies that demonstrate causality, ie that demonstrate that an intervention does have a significant, direct impact on the outcome of choice (ie return to work or work readiness) rather than simply demonstrating an association. Causality can only be demonstrated by research that uses longitudinal study design (ie studies with at least one follow up measure after baseline). Therefore the inclusion criteria were constructed to limit the retrieved studies to those reporting longitudinal data.

The exclusion criteria were as follows:

Excluded locations were:

■ any non UK.

Excluded populations were:

- all adults over age 16 in full or part-time employment, both paid and unpaid
- all adults over age 16 not in receipt of incapacity benefit (or a previous version of the benefit).

Excluded Interventions, programmes, policies and strategies were:

- were delivered outside a workplace or primary care setting, with no primary care or employer involvement in the planning, design, delivery, management or funding
- deal solely with the provision of pharmacological treatment
- looked at the effectiveness of private health insurance schemes, the incapacity benefit system and/or the claiming of statutory sick pay to reduce sickness absence.

Excluded study types:

- studies which describe the relationship between health/ill health and incapacity (ie correlates studies or non evaluative studies of an intervention, policy, programme or strategy). Descriptive studies of participants' views and experiences and cross-sectional studies (ie with only one data collection point) are also excluded.
- dissertations/theses
- non-English language studies.

2.2.2 Development of title and abstract screening checklists

Detailed sifting criteria in the form of title/abstract screening checklists were developed. The inclusion/exclusion criteria outlined in the scope guided the sift process and helped to ensure consistency in screening across the abstract sifting team. In total, the following three abstract screening checklists were developed and used to screen the titles and abstracts retrieved:

- 1. Effectiveness:-primary studies: RCTs and longitudinal intervention studies
- 2. Economics:-primary studies and reviews literature
- 3. Reviews:-reviews of the effectiveness literature.

The abstract screening checklists are given in Appendix 2.

2.2.3 Title and abstract screening process

When a title/abstract appeared to satisfy all of the inclusion criteria it was coded as 'include'. When it was unclear regarding its possible relevance and inclusion, then it was coded as 'get full paper'. Any titles and abstracts meeting any of the exclusion criteria were excluded from the review.

Titles and abstracts coded as 'get full papers' were retrieved because there was insufficient information contained in the title or abstract to assess whether the study was relevant or not. Consequently, when the full paper was acquired, an accurate assessment could be made as to whether or not to include the paper in the review.

To be considered an economic evaluation, a study has to analyse explicitly both the costs and the outcomes of the intervention under investigation in comparison to the costs and outcomes of at least one alternative. For the economic results, abstracts could also provide potentially relevant background data which may assist with the modelling review – any such abstracts were tagged and coded appropriately.

In a reciprocal process, any titles and abstracts from the searches for the two other evidence reviews (which cover research questions 1 to 3) that appeared relevant to research question 4 were also tagged as such and then re-screened using the abstract screening checklist in Appendix 2.

The result was a list of papers coded as 'includes' or 'get full paper' from the effectiveness and cost-effectiveness literature searches for this review (covering research question 4) and subsequent evidence reviews (covering research questions 1 to 3), and the suggestions from experts and the PDG. From each source of references, those passing the title and abstract screening for research question 4 were ordered for full paper retrieval.

2.2.4 Additional website searches

The searches results for the additional five websites (see above section 2.1.4) identified 280 potential papers for question 4 (and 611 for question 1 to 3). The Xpert HR website search yielded a high number of hits (267 for research question 4 and 580 for research questions 1 to 3 prior to de-duplication). However, it was felt that the titles/abstracts identified by these searches may not be particularly relevant and the time required to screen these results would only yield minimal relevant material. In order to determine this a ten per cent random sample of the titles/abstracts were selected and screened using the appropriate forms by the NICE team. Ten per cent of the sample were considered potentially relevant and coded as 'get full paper'. Given the small percentage it was agreed that the remaining titles/abstracts from these search results would not be screened at this stage. This therefore left 13 search results for review question 4 from the remaining four additional websites searched (and 31 search results for research question 1 to 3). None were identified as potentially relevant to research question 4 following title/abstract screening and ordered for retrieval.

2.2.5 Number of papers ordered

Table 2.1 summarises the numbers of titles and abstracts coded as 'include' or 'get full paper' for each of the above categories. It should be noted that because of the reclassification of papers (eg papers from the searches for the two other reviews

Table 2:1: References passing research question 4 title and abstract sift by source				
Source	Number of title and abstracts screened	Number coded as include or get full paper for research question 4		
Q4 Effectiveness primary study search	5,627	118		
Q4 Economics primary study and review search	353 (primary studies)	22		
Q4 Effectiveness review search	716	23		
Q1-3 Effectiveness primary study search	15,345	24		
Q1-3 Economics primary study and review search	2,495	1		
Q1-3 Effectiveness reviews search	309	0		
Q4 Additional website search*	13	0		
Q1-3 Additional website search*	31	0		
Q4 PDG & Expert suggested references	21 (effectiveness)	5 (inc. one review)		
Q3 PDG & Expert suggested references	18 (effectiveness)	0		
Total		194		

* includes four websites suggested by the PDG. It was decided not to fully screen the X-pert HR website (see 2.2.4)

Source: IES/IWP/ScHARR

(covering research questions 1 to 3), which were tagged as a research question 4 paper, then excluded when screened against the research question 4 checklist and the elimination of duplicates) the figures presented below do not necessarily match those in previous reports. In total 194 references were ordered for full paper retrieval.

2.2.6 Full paper screening

Of the 194 papers ordered 20 (effectiveness studies) turned out to be books or book chapters so were excluded from the review. Of the remaining 174 papers ordered items coded as 'includes' or 'get full papers', 12 (effectiveness references) are unavailable (eg because they are incomplete). The rest have been retrieved and screened.

Detailed full paper screening checklists were developed to allow a more accurate assessment of whether each paper met all the inclusion and exclusion criteria (See Appendix 3 for details of the three full paper screening checklists developed). The full paper screening checklists were based on all the inclusion and exclusion criteria detailed in section 2.2.1. Some of these inclusion and exclusion criteria were expanded to reflect the detail in the scope or parameters discussed and agreed by the PDG. This detail enabled a better categorisation of papers included at the full paper screening stage and ensured that any data relevant for the economic modelling was identified and coded appropriately. The expanded criteria are as follows:

Additional or expanded inclusion criteria:

Interventions, programmes, policies and strategies:

- These can be delivered in a primary care setting and/or workplace setting and/or planned, designed, delivered, managed or funded in collaboration with primary care providers and/or employers. These interventions can be delivered by a number of providers (such as voluntary, private and statutory sector) and/or in various settings not just workplace or primary care settings (such as jobcentres, community centres) as long as they are fully or co-planned, designed, delivered and/or funded in collaboration with employers and primary care service.
- Could help adults (over 16) who are unemployed and in receipt of incapacity benefit (or a previous form of incapacity benefit or similar benefit) return to work (paid/unpaid) or prepare for work (paid/unpaid).

Primary outcomes

■ Return to work (paid/unpaid).

- Sustained return to work(paid/unpaid).
- No effect on return to work, job related activity or any other work outcome.

Secondary outcomes

 Other work related outcomes (ie uptake of or increased job seeking, increase in work experience or vocational training, and increase in skills/knowledge for work/unpaid work or alternative career/work)

To be included in the review the paper had to contain data on the primary review question. If a paper only addressed secondary research outcomes it would be excluded.

Study design for effectiveness papers:

At the full paper screening stage the study design of all included papers was classified into one of the following categories to allow grouping of papers for data extraction and quality assessment:

- RCT
- controlled before and after
- cohort
- case control
- before and after
- interrupted time series
- other.

Study design for cost effectiveness papers:

At the full paper screening stage the study design of all included papers was classified into one of the following categories to allow grouping of papers for data extraction and quality assessment:

- cost benefit analysis (CBA)
- cost effectiveness
- cost utility
- other.

14 Interventions to Help Recipients of Incapacity Benefits Back to Work

Mixed studies

 If the study was a combined intervention and the public health data could be disaggregated.

Mixed studies are ones that included data on treatment and an intervention, policy, strategy or programme relevant to this review or covered multiple population groups which met both inclusion and exclusion criteria. Mixed studies would only be included if the data relevant to the appropriate population group could be sufficiently disaggregated.

Additional or expanded exclusion criteria:

Interventions, programmes, policies, strategies:

- Studies which deal solely with the effectiveness of the incapacity benefit system, private health insurance schemes or statutory or occupational sick pay.
- Studies dealing solely with preventing ill health retirement (ie where the individual has no intention of returning to work).

Study types excluded:

- Studies which describe the relationship between health/ill-health and incapacity benefit (ie correlates studies or non evaluative studies of an intervention, policy, programme or strategy).
- Books and book chapters

2.2.7 Review-level material

As noted above in Section 2.1, searches were also made for relevant review-level material, such as effectiveness and cost effectiveness reviews or meta-analyses of RCT's and longitudinal studies. A further review-level reference was also suggested by PDG and expert – see Table 2.1. All 24 review-level references (23 effectiveness and 1 economic review) assessed to be potentially relevant to the incapacity review were also ordered for full paper retrieval – see table 2.2. A full paper screening checklist was developed and used to screen these reviews (See Appendix 3).

Additional details on the screening criteria for review studies were also specified as follows:

Population inclusion criteria

The review must wholly or partly cover evaluations of an intervention/policy/ strategy/programme which aims to help adults over 16 who are unemployed and in receipt of incapacity benefit (or a previous form of incapacity benefit or other similar benefit) return to work (paid or unpaid).

Setting inclusion criteria

At least one of the studies reported in the review must be set in the UK

Effectiveness study design criteria:

• The review must include at least one RCT or longitudinal study.

Cost effectiveness study design criteria:

 The study must contain effectiveness studies or economic evaluations with cost effectiveness, cost benefit, cost utility, cost consequences, cost minimization or net monetary (cost) benefit data.

If a review met the full paper screening inclusion and exclusion criteria its reference lists were then checked by two reviewers to identify potentially relevant additional primary studies. Any duplicates with primary study references already obtained were removed and titles/abstracts ordered for retrieval. Abstracts of any primary studies thus identified were then screened using the appropriate abstract screening checklist and if accepted/included, they were added to the references requested for full paper retrieval. As Table 2.2 indicates no additional primary studies were identified from the three effectiveness and one cost-effectiveness reviews screened (none outstanding).

	Review studies ordered	Review studies screened and included in reference tracking	Number of primary studies identified from reference checking of relevant reviews.
Q4 Effectiveness Reviews for reference checking	23	3	0
Q4 Cost-Effectiveness Reviews for reference checking	1	1	0
Source: IES/IWP/ScHARR			

2.2.8 Reference tracking

Additional to the reference checking of included review-level studies, the reference lists for all primary studies that met the inclusion criteria were examined to identify any additional relevant references. A list of any additional references thought to be relevant was checked against the Reference Manager databases of literature search results. Abstracts were obtained for any references not identified in the Reference Manager databases (previously identified references would have already been screened) and they were then screened following the same process described above for titles and abstracts generated by the search of electronic databases. Eleven further, additional effectiveness references were identified by this process and ordered for full paper screening. None were included in this review as a result of this exercise.

2.2.9 Citation searching

The citations of all three included papers were also searched for by one reviewer using Web of Science to determine whether any additional papers citing these included references had been missed. No further, additional references were identified by this process.

2.3 Summary of studies identified for inclusion

The sifting of the titles and abstracts produced by the searching of the electronic databases gave a total of 180 primary papers that possibly matched the inclusion criteria for review Question four. The full papers for these were requested and all have been received and a full paper screening undertaken.

2.3.1 Effectiveness studies

As Table 2.3 outlines 157 effectiveness primary studies were thought to be relevant to research question 4 and were screened. One reviewer checked all of the received papers against the inclusion criteria using the full paper screening list checklist given in Appendix 3 and at least one in ten were double checked by a second reviewer. As a result of this process three papers were included (two from the question 4 effectiveness search and one from suggestions from external experts).For the sources of all possible studies to answer this question, and the points at which studies were excluded, see Figure 2.1.

	Full papers ordered ¹	Full papers included
Q4 Effectiveness primary study search	118	2
Q4 Cost-Effectiveness search	22	0
Q 1-3 Effectiveness primary study search	24	0
Q1-3 Cost-Effectiveness search	1	0
Q4 Additional website searches	0	0
Q4 References suggested by PDG and experts- effectiveness	4*	1
Q1-3 References suggested by PDG and experts- cost effectiveness	0	0
Reference tracking of included papers - effectiveness	11	0
Reference tracking of included papers - cost effectiveness	0	0
Citation searching of included papers (effectiveness)	0	0
Citation searching of included papers (cost- effectiveness)	0	0
Total	180 (23 cost effectivenes and 157 effectiveness)	ss 3 (effectiveness)

Table 2.3: Papers ordered, pending and included

* One PDG suggestions was a review (see Table 2.2)

Source: IES/IWP/ScHARR

The following three effectiveness papers were identified for inclusion using the above process and went forward for data extraction and quality assessment:

- Bewley H, Dorsett R, Haile G (2007) The Impact of Pathways to Work, DWP Research Report 435
- Desouza M, Sycamore M, Little S and Kirker S G B, (2007) 'The Papworth early rehabilitation programme: Vocational outcomes', Disability and Rehabilitation, April 2007, 29 (8), p 671-677.
- Winspear D, 'Using CBT to improve employment outcomes for incapacity benefit customers: Interim report', Journal of Occupational Psychology, Employment and Disability, Vol 9, No 1, Spring 2007, p 41-51.

Appendix 4 provides the reference details of the primary studies excluded as a result of the full paper screening. Studies were excluded because they failed to meet at least one of the inclusion criteria (or met an exclusion criterion). As soon as they failed to meet one of the criteria they were excluded. In the appendix the references are

¹ This includes references which turned out to be books or unobtainable and therefore unable to be screened

ordered by the criterion by which they were excluded. Most were excluded because they were a non-intervention study (did not examine a interventions, strategies, programmes or policies to help incapacity benefit recipients return to work (47 per cent), but these may well have failed against other criteria too. The other main reasons for the exclusion of the studies were that: they were not conducted in the UK (31 per cent); they were based on the wrong population (eg included people employed or not on Incapacity Benefit) (11 per cent); or they were not the right study design (ie RCT or longitudinal studies) (five per cent); or did not provide data on the primary outcomes for this review (five per cent).

2.3.2 Cost-effectiveness

When sifting the list of titles and abstracts generated by the search of electronic databases, 23¹ primary studies were selected as possible papers that satisfied the inclusion criteria for this review. One primary study was a duplicate found by two of the searches (Leon et al 2002), so, the final total number of primary studies before screening of full papers was 22. Two reviewers then assessed these full papers using the economic full paper screening checklist. However, when checking the inclusion criteria against the full details of the study (as described in section 2.2), all 22 studies were ultimately excluded because none satisfied the inclusion criteria. For the sources of all possible papers to answer this question, and the points at which papers were excluded, see Figure 2.2.

In the absence of any relevant papers satisfying the inclusion criteria, it was agreed to check other potential sources of possible data, i.e. books and book chapters, which had been tagged during screening of the literature search results. Eleven books and book chapters had been tagged in this way, but, once again, no economic studies with cost effectiveness, cost benefit, cost utility, cost minimisation or net monetary (cost) benefit data were identified for inclusion. The papers and the reasons for all the exclusion are presented in section 4.1

Seven papers were also identified which contained potentially relevant data for the modelling stage. However, this figure may well change once it had been agreed what areas will be modelled.

Figures 2.1 and 2.2 below present a diagrammatic overview of the results of this comprehensive searching and screening process.

¹ Twenty two primary studies from the research question 4 searches and one study from the search for research questions 1 to3.



Figure 2.1: Process for identifying effectiveness studies



2.4 Data extraction and quality appraisal

2.4.1 Data extraction

The study type of each included effectiveness paper was identified using the following algorithm which was adapted from *Methods for development of NICE public health guidance* (Figure 2.3).



Figure 2.3: Algorithm for classifying primary study designs about effectiveness

The effectiveness data extraction form contained in the *Methods for development of NICE public health guidance* was adapted to reflect the parameters of this review – please see Appendix 6 for a example of a completed form. One reviewer extracted data for each full paper using this form. A second independent reviewer checked the data extraction, and any differences were resolved by discussion with a third reviewer.

For the cost effectiveness review, the data extraction form contained in the *Methods for development of NICE public health guidance* was adapted to reflect the parameters of this review and supplemented with questions from the Drummond checklist (Guidelines for authors and peer reviewers of economic submissions to the BMJ, M F Drummond, on behalf of the BMJ Economic Evaluation Working Party) – please see Appendix 7.

Secondary outcomes

If a study is included on the basis that it contains data relevant to the primary research question and outcome, then data on any secondary outcomes will also be reported. The following secondary outcomes are to be considered (but are not limited to):

- Other work related outcome (eg increase in work experience, vocational training, increase in knowledge/skills for alternative work/career, uptake and increase in job seeking activity)
- Acceptability of the intervention/policy/programme/strategy content/frequency/location etc.
- Identification of any adverse or unintended (positive or negative) outcomes as a result of the intervention, programme policy or strategy.
- Barriers to or facilitators of effective implementation
- Individual improvement in personal aspects such as: ability to cope, mental condition, disability management, changed capacity and goals, musculoskeletal ability and social functioning.

2.4.2 Quality assessment

Effectiveness studies

Quality appraisal for the effectiveness studies was conducted based on the NICE CPHE forms. These forms provide criteria for rating a study based on how robust an example it is of that particular study design. For example, a randomised control trial (RCT) will be rated on how well it meets the defined standards for a robust RCT. Different criteria exist for each type of study design. This means that the quality rating for studies of the same design can be compared with each other (ie an RCT rated ++ is more robust than an RCT rated +). However, quality ratings for different study designs cannot be compared and an RCT rated – is still likely to be provide more robust data than a before and after study rated ++ because an RCT is an inherently stronger study design.

It was agreed that the criteria for making the actual quality assessment ratings (++, +, -) should be adapted to reflect study designs found in the social sciences/public health area (eg an RCT would not be downgraded for failure to use complex concealment designs such as double blinding), because such designs are not always possible with intervention studies where the individual delivering the intervention knows what the intervention is.

Two independent reviewers assessed the quality of each included study. Any differences in quality assessment were resolved by discussion with a third reviewer
or, if agreement could not be reached, details were reported in the review. Appendix 6 gives the quality assessment forms used and a completed checklist is given.

The quality assessment checklists contained in the *Methods for development of NICE public health guidance* were adapted to reflect the parameters of this review – see Appendix 7.

Cost effectiveness studies

No economic quality assessment was undertaken given no economic studies were identified for inclusion.

2.5 Synthesis and formulation of evidence statements

2.5.1 Effectiveness studies

The results of the data extraction and quality assessment for each theme identified in the included effectiveness studies were presented in a narrative summary and combined in a summary evidence table. An evidence statement was generated for each theme. Chapter three of the report presents the synthesis of data and evidence statements for the included effectiveness studies.

2.5.2 Cost-effectiveness studies

No synthesis was undertaken of cost-effectiveness studies because no study satisfied the required inclusion criteria.

3 Effectiveness Findings

Three effectiveness studies were identified which met the criteria for inclusion for interventions, programmes and strategies to help recipients of incapacity benefits return to work. The interventions described differ considerably and are discussed under the three separate thematic headings below.

The evidence tables for each of the studies are presented in full alphabetical order (by the first named author) at the end of this chapter.

3.1 Theme 1: Cognitive behavioural therapy

Winspear (2007)

A Randomised Controlled Trial (RCT) study (rated '-') conducted in 2007 investigated the impact of cognitive behavioural therapy (CBT) delivered by Jobcentre Plus work psychologists over an eight week period to improve the employment outcomes for incapacity benefit recipients with mild, moderate or severe anxiety and depression (an average 8.7 sessions per individual). The published paper presents an interim report on the research¹.

The intervention was delivered by ten Jobcentre Plus work psychologists all of whom had attended a five day CBT course (one already held a qualification in CBT). The participants were 67 incapacity benefit recipients with mild, moderate or severe anxiety and depression. No details are given in the paper of their demographics or employment histories. The intervention is being trialled in two demonstration sites in Newham and Doncaster. The participants were referred to the study by Incapacity Benefit Personal Advisers and Disability Employment Advisers using the criteria of mild/moderate levels of anxiety/depression, a desire to return to work but finding it

¹ Data on the final outcomes have been requested but are yet to be available

difficult because of mental health problems and a willingness to participate in a psychological therapy. Those referred were randomly assigned to either a group for immediate therapy or to a control group which waited eight weeks to commence the programme. No information was given on how random allocation was achieved (ie who did it, was it blind, etc.)

Those taking part were assessed on a number of measures of psychological health, employability and job-seeking behaviours at the start of the intervention, post intervention and at a three month follow-up stage. Additionally the control group was assessed at the start of the waiting period of eight weeks. The published standardised scales of Rosenburg Self Esteem, Beck Depression Inventory and Beck Anxiety Inventory were used. An Approach to Work scale and a Job-Seeking Behaviours scale designed by the author were also used but it is not stated if these questionnaires had been validated. The employment status of the participants was gathered at a three month follow-up.

The study is a progress report on the research using partial data, a point that was stressed by the author. Ten of the 67 people recruited to the research were deemed unsuitable because they had a range of problems outside the selection criteria such as drug or alcohol problems, traumatic brain injury or long-enduring mental problems. A further 17 dropped out and the reasons for this are not known by the author. Eighteen participants had completed the course and the remaining 22 people were still completing their eight weeks of CBT.

There were some notable discrepancies in the reporting of the findings section of the paper. The text and Table 4 of the paper refer to 14 out of the experimental group and four out of the control group as having completed the programme, however, these figures are transposed in Table 2 of the paper. Table 2 in the paper also indicates that 18 participants have completed their CBT but the text refers to 35 subjects who have finished their CBT intervention. 'Finished' is therefore assumed to refer to both the 18 completers and the 17 from both the experimental and control groups who dropped out of the programme.

The findings with regard to base line depression and anxiety scores also appear to be based on the sub-sample of 14 of the experimental group and four of the control group who have completed their CBT intervention plus the ten from the experimental group and the seven from the control group who have dropped out. It is reported that just over half of these participants reported high levels of depression and or anxiety at baseline but no further information is available about whether they were more likely to complete or leave the intervention.

Changes in depression anxiety and self esteem were assessed by comparing control group (n= 11) pre-intervention mean scores with intervention group (n=14) post intervention mean scores. These show significantly better scores in the intervention group for depression (p=0.000), self-esteem (0.016) and anxiety (p=0.019).

Primary outcome:

The interim results showed that four of the 57 participants allocated to receive CBT had found work but no further details are given and one had been referred to a job broker for help with securing a job.

Secondary outcomes:

The secondary outcome in the study of relevance to this review relates to improved employability. This was measured via ability to relate to others, work motivation, proactive job searching and control over job search. Frequency of job seeking behaviour was also assessed.

Results for the experimental group alone pre and post-intervention show significant increases in participants' control over their efforts to find work (p=0.000), their motivation towards work (p=0.009), their ability to relate to others (p=0.028) and their proactivity in job seeking (p=0.036). However, it is unclear whether these figures relate to all in the experimental group or just to those completing the intervention as no base numbers are given in the table presenting the mean scores.

No differences in actual job-seeking behaviour were found between the preintervention control group and the post-intervention experimental group immediately after finishing the CBT intervention. The author comments that is more realistic to expect to see increased job-seeking behaviours at the three month follow-up stage.

The paper presents no data concerning any of the secondary research questions listed in Section 1.2 of the introduction.

Limitations of the study

A number of limitations of the study were identified:

- The experience of CBT of those delivering the programme was limited and potential differences between them were not tested.
- No details of the intervention content are given.
- No reasons for the varying numbers of sessions among the participants are given.
- The sample of 57 individuals is small and details of the random allocation were not presented.
- Characteristics of the experimental and control groups in terms of gender, age or any other features were not given so the extent to which they were matched is not known.

- Two of the outcome measures were developed by the author and it is not known if these were validated.
- Differences between completers and those dropping out of the programme are not analysed.

Owing to these limitations of study design, both reviewers assessed this RCT study as '-'.

Evidence statement:

ER3.1: There is insufficient evidence from one RCT (grade -) to assess the efficacy of cognitive behavioural therapy delivered by Jobcentre Plus work psychologists over an eight week period in improving employment outcomes for incapacity benefit recipients with mild, moderate or severe anxiety and depression and expressing a desire to return to work (Winspear 2007).

3.2 Theme 2: Work-focussed interviews and access to employability support

Bewley et al. (2007)

This study (rated '+') is a non-randomised area controlled trial based on a before and after comparison between areas that had implemented the Pathways to Work programme and similar areas which had not. The Pathways to Work pilot programme aimed to encourage employment among people claiming incapacity benefits through compulsory attendance at a work-focussed interview and access to return to work support.

In October 2003 the Pathways to Work initiative was launched in three Jobcentre Plus (JCP) districts in England, Wales and Scotland. Six months later, April 2004, pilots started in four more areas in England. The Pathways programme, at that time, involved a number of elements. Individuals aged between 18 and 60 not in work and making a claim for incapacity benefit (IB) were required to attend a Work Focussed Interview (WFI) with a trained IB Personal Adviser (IBPA) eight weeks after making their claim. Failure to attend the interview could have led to benefit sanctions. Claimants had to attend a further five WFIs unless they were judged either to have particularly severe medical conditions or likely to return to work without further help.

Participation in all other aspects of Pathways was voluntary and could involve the following elements:

 a 'choices' programme of training or support to help people enter the labour market, including the New Deal for Disabled People (NDDP) and a Condition Management Programme (run with local health providers to help people manage their health condition);

- a Return To Work Credit (RTWC) of £40 a week (for up to a year) for working over 16 hours a week in a job earning less than £15,000 pa;
- In-Work Support (IWS) including one or more of the following: mentoring, a job coach, occupational health support, financial advice and in-depth support to complement that provided by IBPAs and NDDP job brokers [no other information about this support is provided in the study];
- Advisers' Discretionary Fund (ADF) to make purchases of up to £100 to help people find work.

The overall evaluation of the Pathways programme has a number of strands. This impact evaluation study examines the overall effect of the programme (and does not look at the component parts). It involved analysis of two streams of quantitative data for each of the two sets of pilot areas (those starting in October 2003 and those in April 2004):

- administrative data on the 23,300 claimants in the pilot areas during the study period (from the National Benefits Database) including personal characteristics and benefits claims history for each of 18 months after the initial benefit enquiry;
- data from telephone surveys of individuals who had started the claims process (although two slightly difference sample frames were used one covering people who had made an initial enquiry and the other of those who had actually started making a claim). Surveys (of separate samples) were conducted soon after individuals made their first enquiry (baseline) and around 19 months after the initial enquiry. The sample sizes for the October 2003 and April 2004 pilots (ie before and after,) were 1794 (in both October 2003 and April 2004 pilot areas) before Pathways started and 1957 (October and April areas) at the follow-up point (details of response rates etc. were not provided in this report).

Equivalent administrative and survey data were also collected from comparable Jobcentre plus areas that had not yet implemented the Pathways to Work programme and which formed a control or comparison sample (the survey samples for the control areas were 723 (at the 'before' point) and 1310 (at the 'after' point)). No comparisons on the pilot and control samples are provided in this study report.

Primary outcomes

Average employment levels among those receiving the Pathways to work Programme were higher, but not significantly, than the equivalent people in the control areas and therefore not receiving the programme 18 months after their initial IB enquiry.

The impact analysis was conducted using a 'difference-in-difference' (DiD) approach which compares the difference between employment rates before and after the introduction of Pathways in the pilots areas with employment rates at the same points in the control areas using a regression framework to control for observable differences between the two samples.(The variables used in the DiD analysis of the survey data were: sex; age; dependent children; ethnicity; age left school; whether had a partner and partner's employment status; type of health problem experienced; length of time health problem lasted).

Using the survey data the authors found that employment levels for those receiving Pathways to Work were 7.4 percentage points above the estimated 'base' of 29.7 per cent who did not receive the programme (p = 0.09))¹. In other words, in the Pathways areas 37.1 per cent of those making an IB claim were in work 18 months on, compared with assumption that 29.7 per cent (based on data from the control areas) would have been in work if Pathways had not been operating. This employment effect was reported to be 'quite stable over the latest six or so months observable'. There was a smaller positive, but not significant, effect on employment of 16 hours a week or more (p = 0.18) and in paid work over 30 hours or more (p = 0.40) in people receiving the programme compared with those not receiving the programme.

The effect on earnings was positive (but not statistically significant (p = 0.40)).

There was a small positive effect at the 18 month point on the receipt of incapacity benefits (ie a reduction of 1.7 percentage points on a base of 51 per cent) but this was also statistically not significant (p = 0.72). Administrative data indicated that there was a larger effect in the first six months after making a claim which declined over time.

Overall these results indicate that Pathways increased the likelihood of working among those making a claim for Incapacity Benefits a year and half after the original IB enquiry.

Secondary outcomes

Participants were significantly less likely to report self-assessed health problems which affected day-to-day activity 'a great deal' – by an average 10.8 percentage points from the 49.8 per cent 'base' estimated from the control group (p = 0.02).

Sub-group analysis² showed that Pathways had:

¹ This is statistically significant at 90 per cent but not at 95 per cent confidence limits

² The analysis at sub-group level does not control for compositional differences between the subgroups which may influence any observed effects. Thus, for example, if the employment effect is greater for women than for men it cannot be inferred that being female is likely to increase the effect

- stronger employment effects among women compared with men (the employment effect¹ for women (n = 1505, pilot and control sample combined) was an average of 13 percentage points, a statistically significant result comparing the pilots and the controls (p = 0.05), and compares with a three percentage point increase for men (n = 1786), which was not statistically significant (p = 0.62);
- stronger employment effects on those aged under rather than over 50 (employment among the under 50's (n = 2100) was 10.6 percentage points higher that it would have been without Pathways (p = 0.06)² compared with only 2.3 percentage points higher for the over 50's (n = 1190), (p = 0.75);
- little effect on the employment of those whose main health condition at first interview involved mental illness (employment levels among those with a mental illness (n = 700) was estimated to be -1.1 percentage points (ie lower than it would have been if Pathways had not been operating) (p = 0.9), compared with a 10.7 percentage points increase for those with no mental illness (n = 1985) (p = 0.06).

Limitations of the study

The study has been quality assessed and given a rating of '+'.

The authors state that the survey sample population was drawn from a different sample source to the administrative data (the former were those making an initial enquiry about and IB claim and the latter had actually started making a claim) and that this could have affected some of the results.

Of the seven pilot areas, three started Pathways in October 2003 and four in April 2004. During this time, considerable organisational change took place in the Jobcentre Plus offices as a new regime was introduced on an area by area basis. The Jobcentre areas involved in the October 2003 pilots were among those in the first wave of these changes and could have been disproportionately affected (eg in their ability to deal with claimants) compared to their control areas, some of which underwent the changes at a later date. For this and other related reasons most of the reported analysis was based on the latter areas.

of Pathways; rather, it is the case that the combined characteristics of women predispose them to being affected more by Pathways than men.

¹ Ie the estimated level of increase in the proportion employed above what it would have been in the absence of the programme (ie the counterfactual estimated from the control areas)

² ²This is statistically significant at 90 per cent but not at 95 per cent confidence limits

Evidence statement:

ER3.2: There is limited evidence from a non-randomised controlled trial (grade +) that a programme comprising attendance at a work-focussed interview and access to return to work support (including further interviews, employability training, help with managing their health condition, financial support and in-work occupational health and personal support) could be effective at increasing the chances of people on Incapacity Benefit (IB) being in work 18 months after initially enquiring about accessing IB. The employment effects appear to be stronger for women than men, those aged under, rather than over, 50 and people without rather than with mental illness. (Bewley et al. 2007)

3.3 Theme 3: Rehabilitation programme

Desouza et al. (2007)

A case series study; grade '-'evaluating the Papworth Trust's Early Rehabilitation Programme was conducted in 2007.

The programme, based in Eastern England, aims to provide those with physical or cognitive disabilities with 'employment, medical and life skills' to enable them to return to work and is open to anyone receiving incapacity benefit. The programme provides support for participants over a four to ten month period from a rehabilitation coordinator, case manager, job coach and assistant, vocational adviser, information technology assessor and vocational psychologist. Help from a literacy tutor, an occupational therapist and assistant, a consultant in rehabilitative medicine, a speech and language therapist and a physiotherapist is also used according to individual need. Clients are often enrolled in local adult education courses, vocational training and certification programmes aimed at the general population.

The study involved reviewing all the records of the 340 individuals who had contacted the Papworth Trust (a UK charity promoting equality, choice and independence for people with physical disabilities, sensory impairment and learning difficulties) between 1995 and December 2003 and had completed referral forms to assess their eligibility for the programme. This exercise identified 107 individuals who had started the rehabilitation programme with 94 completing it. The reasons for exclusion given in the paper were the non-attendance of the next recommended stage of the acceptance process (132), those who were not recommended for acceptance (73), those who left the programme before completion (13) and those who could not proceed as no funding was available (20). It should be noted that these stated reasons account for five more people than the total number of clients.

The median time for contact to be made with those completing the programme was 48 months after completion. No control group was established. The participants

comprised 87 males and 20 females with the following conditions: brain injuries (54 per cent), musculo-skeletal injuries (21 per cent), back injuries (14 per cent) and other injuries (11 per cent). All participants were contacted by phone to determine their vocational position at November 2004 but their responses were not validated.

Primary outcome:

The results showed that of the 94 who completed the programme at the time of follow up, 53 were employed, and two were in voluntary work. The remaining 39 were unemployed.

Secondary outcomes

Thirty-three participants were deemed 'work ready' when contacted. Work readiness was defined as having sufficient stamina, dexterity, basic cognitive skills and interpersonal skills to be able to seek and sustain employment with the help of schemes run by Disability Employment and Job Centres. It is unclear from the paper if just some of the participants undertake a work placement or if this is an integral part of the programme. However, at the end of these work placements those taking part are deemed 'work ready' and are given a written reference from their manager. The measures for assessing work readiness are also not reported in the paper.

The remaining six participants in the programme were unemployed but four were in education, although no further details are given in the paper about this, and two had achieved independent living which is one of the objectives of the programme.

When their conditions were considered those with back injuries and musculo-skeletal injuries appeared more likely to have returned to work (70 per cent) than those with brain injuries (34 per cent), however the base numbers are small, 37 for back and musculo-skeletal injuries and 58 for brain injuries.

This paper also presents no data concerning any of the secondary research questions listed in Section 1.2 of the introduction.

Limitations of the study:

The following limitations were identified in the study:

- The effectiveness of the intervention can not reliably be assessed in the absence of a control group.
- The effectiveness of the programme is evaluated after a different length of time for each of the participants rather than a fixed interval. The results therefore include those who have only just completed the programme along with those who took part nine years ago.

 A high number of people were excluded, ineligible or dropped out which produced a very selective sample (potentially self-selecting sample) which limits the generalisability of this intervention.

For these reasons the paper has been coded as a case series study; grade '-'.

Evidence statement:

ER3.3: There is insufficient evidence from one case series study (grade -) to assess the efficacy of the Papworth Trust's Early Rehabilitation Programme (comprising support for participants over a four to ten month period from a rehabilitation coordinator, case manager, job coach and assistant, vocational adviser, information technology assessor and vocational psychologist - help from a literacy tutor, an occupational therapist and assistant, a consultant in rehabilitative medicine, a speech and language therapist and a physiotherapist are also used according to individual need) in assisting the return to work of those on incapacity benefit because of disabling injuries (Desouza et al 2007).

3.4 Evidence tables for the two themes

Table 3.1 presents a summary of the evidence from the three included papers.

Table 3.1: Evidence tables for the studies included in the effectiveness review

Bewley et al. (2007)

Study details	Intervention, policy, strategy or programme description	Sample and setting	Duration of study and follow-up period/s	Primary and secondary outcomes	Results	Confounders and limitations
Title and source:	Aim:	Included:	Duration and	Primary outcomes	Primary outcomes	Identified by
Bewley H, Dorsett R, Haile G	To evaluate the effect of the Pathways to Work	23,300 people in the seven pilot	follow-up: Administrative data were	Return to work (in employment (at 16	18 months after their initial IB enquiry, the employment levels in	author: The survey sample
The impact of Pathways to Work,	aimed to encourage employment among	areas who made an enquiry about and/or started to	collected from the sample 18	+ or 30 + nrs pw) 18 months after their initial claim.	the eligible population were estimated to be 7.4 percentage	population was drawn from a different sample
DWP Research Report 435, 2007	people claiming incapacity benefits through compulsory attendance at a work-	make a claim for IB between 1 August and 30 November 2004.	their initial enquiry. These data were	Receipt of incapacity benefit, 18 months after	points higher above the 29.7 per cent of individuals who would have been in work if Pathways had not been operating (p =	source to the administrative data (the former were those making an
Study design:	focussed interview and	Findings based on	supplemented by survey data	their initial claim (and monthly	0.09).	initial enquiry
Before and after comparison study	access to return to work support.	administrative data for entire	from two samples of a)	during the intervening period.	There was a smaller positive, but not significant, effect on	about and IB claim and the latter had actually started
OA Grade: +	Intervention:	sample surveys of	1794 enquirers and claimants	Earnings, 18	employment of 16 hours a week or more ($p = 0.18$) and in paid	making a claim).
	Individuals aged between 18 and 60	the population (and controls)	in the pilot areas before	months after their initial claim.	work over 30 hours or more ($p = 0$ 4).	Of the seven pilot areas, three started
	making a claim for incapacity benefit (IB)	Excluded:	Pathways started and b)	Secondary	The effect on earnings was	Pathways in October 2002 and
	were required to attend	Those that were	1957 IB	outcomes	positive (but statistically insignificant $p = 0.4$)	four in April 2004.
	Interview (WFI) with a	have particularly	enquirers and claimants in the	health condition	There was a small positive effect	Due to the timing of its
	trained IB Personal Adviser (IBPA) eight weeks after making	severe medical conditions or likely to return to	pilot areas after Pathways		on receipt of incapacity benefits (ie a reduction of 1.7 percentage	implementation, changes in the
	their claim. Claimants had to attend five further WFIs, unless they were judged either to have particularly	work without further help were excluded from the programme. (It is not clear	started, compared with samples of a) 723 and b) 1310 in the		Sounts on a base of 51 per cent) out this was also statistically nsignificant (p = 0.72). Administrative data indicated that there was a larger effect in	have had a greater impact on the October 2003 areas relative to its

severe medical conditions or likely to return to work without further help. Other aspects of the programme were voluntary and could involve:

- access to a 'choices' programme of training or support to help people enter the labour market:
- a Condition Management Programme (run with local health providers to help people manage their health condition):
- Return To Work Credit (RTWC) - of £40 pw for jobs >16hrs pw earning less than £15,000 pa;
- In-Work Support (IWS) including mentoring, occupational health support, financial advice:
- Advisers' Discretionary Fund ADF) - to make purchases of up to £100 to help people find work.

Comparison:

Benefits claimants in

comparison were similarly areas over the excluded from same time the surveys in the periods. pilot and control

Setting:

areas)

Seven Jobcentre Plus districts, one in Scotland, one in Wales and five in England

whether they

the first six months after making comparator areas a claim which declined over time

These results indicate that Pathways increased the likelihood of working among those not receiving benefits a year and half after the original IB enquiry

Secondary outcomes

Participants were significantly less likely to report self-assessed health problems which affected day-to-day activity 'a great deal' by 10.8 ppts from a base of 49.8 per cent (p = 0.02).

Sub-group analysis showed that Pathways had:

stronger employment effects on women rather than men (the impact estimate for women was13 ppts p = 0.05, compared with 3 ppts for men, p = 0.62;

stronger employment effects on those aged under rather than over 50 (the impact estimate for <50's was10.6 ppts p = 0.06, compared with 2.3 ppts for >50's, p = 0.75);

little effect on the employment of those whose main health condition at first interview involved mental illness (the impact estimate for those with a mental illness was-1.1ppts p = 0.9, compared with 10.7 ppts for those with no mental illness, p = 0.06).

than the April 2004. For this and other related reasons most of the reported analysis was based on the latter areas

Identified by reviewer:

none

36 Interventions to Help Recipients of Incapacity Benefits Back to Work

comparable non-pilot areas

Desouza et al. (2007)

Study details	Intervention, policy, strategy or programme description and comparison	Sample and setting	Duration of study and follow-up period/s	Primary and secondary outcomes	Results	Confounders and limitations
Title and	Intervention:	Sample: 107	Duration:	Primary	Primary outcomes:	Identified by author:
source:	Evaluation of	patients on incapacity benefit	Programme	outcomes:	Of the 94 completing the	High drop-out rate 13 of
M. Desouza, M.	Papworth Trust Early Rehabilitation	starting	carried out between 1995	Self-reported return to work.	programme at follow-up there were:	107 starting programme
Sycamore, S. Little &	Programme:- Delivered by	between 1995 and	and 2003.	Unvalidated.	56% (53) employed	Identified by reviewers:
S.G.B. Kirker- 2007	Papworth Trust,	2003. 87 Male, 20 female.	Duration is 4-10 months	Self- reported "ready to	31% (33) work ready	Lack of a control group
Ctudy	occupational	41 traumatic brain	dependent upon patient.	work". Unvalidated. Secondary outcomes:	4% (4) in education,	means effectiveness
design:	therapy and modical	injury, 22 museule skeletel			2% (2) in voluntary work.	cannot be reliably assessed
Case series	rehabilitation.	Injuries, 17 acquired brain	Follow-up:		2% (2) achieved independent living	Effectiveness of the
study	Comparison:	injury, 15 back	Follow-up in		Completers: % of those with	after differing lengths of
Crado	None	injuries, 8 amputees and 4	2004 (median	None reported	condition achieving employment	time from just completed
Giaue		other.	up 48 months	п рарег	traumatic brain injury 32% (13)	
		Setting:	after completion of		acquired brain injury 41% (7)	were excluded, ineligible
		Eastern England	the programme).		musculo-skeletal injuries 64% (14)	or dropped out producing a selective sample
		carried out in			back injuries 80% (12)	
		centres and evaluated			amputation 62% (5)	
		subsequently by			Other 50% (2)	
		follow-up in 2004.			Secondary outcomes:	
					None reported in the paper.	

Winspear (2007)

Study details	Intervention, policy, strategy or programme description and comparison	Sample and setting	Duration of study and follow-up period/s	Primary and secondary outcomes	Results	Confounders and limitations
Title and	Intervention:	Sample: 57	Duration:	Primary outcomes:	INTERIM FINDINGS ONLY	Identified by author:
source:	Cognitive	Incapacity Benefit	Eight week	Return to work (self	Primary outcomes:	Interim findings only.
Winspear D, 'Using CBT to improve employment outcomes for incapacity benefit	Behavioural Therapy (CBT)	recipients with	programme.	reported)	The interim results showed that	
	delivered by	or severe	F - U	Secondary outcomes:	four of the 57 participants allocated to receive CBT had	Identified by reviewers:
	vork psychologists over an 8 week period, with a minimum	anxiety and depression desiring to return to work.	post- programme	Psychological health (Beck's Depression and Anxiety Inventories, Rosenburg Self	found work but no further details are given and one had been referred to a job broker	CBT deliverers trained to different levels; short-term training only
customers:			assessment		for help with securing a job.	No details of intervention content
report',	maximum of 18	demographic or	plus	employability	Secondary outcomes:	No reasons given for varying
Journal of Occupational	sessions (mean = 8.7).	employment details given	3month follow-up.	(Approach to Work Questionnaire,	Compared with baseline/pre-CBT levels: CBT reduces depression (p=0.000) and anxiety (p=0.019) and improves self-esteem. Secondary outcomes: CBT completers (both intervention and control groups) reported improved employability on 4 dimensions (ability to relate to	numbers of sessions among participants
Psychology,	Comparison:	Experimental		comprising 4 dimensions: relating		Very small sample
Employment and Disability, Vol 9, No 1,	No CBT - waiting list controls (start of treatment programme delayed by 8 works Authors	group = 31, control group = 26	group = 31, control group = 26 (NB. Figures wrong way cound in table	to others, work motivation, proactive job searching, control over job search [Winspear 1998]); Eroguency of colf		Details of random allocation not specified
Spring 2007, p 41-51		(NB. Figures wrong way round in table				Extent to which control and experimental group were matched is unknown
Randomised	also compare	2 in the paper).		reported job seeking	motivation (p=0.009), proactive	Not known if two outcome
controlled trial (individual)	experimental group before and	Setting:		penaviour (factual guestionnaire)	job searching (($p=0.036$), control over job search ($p=0.000$)) and	measures were validated
Grade: -	after CBT.	Newham and Doncaster			sought information about jobs more often. Four found work.	Differences between completers and those dropping out were not analysed

3.5 Applicability of the evidence to the UK populations in the scope

Three studies have been included in the effectiveness review; one describing the use of CBT sessions with those on incapacity benefit to improve vocational outcomes (Winspear, 2007), one evaluating a rehabilitation programme to aid return to work for those with disabling injuries (Desouza et al. 2007) and the third was a major evaluation of a change in the level and nature of support provided to Incapacity Benefit claimants piloted in areas throughout Great Britain (Bewley et al. 2007).

The Winspear study had a number of limitations which led to its '-' rating. No details were given of the intervention content, the method of random allocation, the varying number of sessions among the participants or the characteristics of the experimental and control groups to assess the extent to which they were matched. In addition the experience of CBT of those delivering the programme was limited and the potential differences between them were not tested. Any differences between those completing the programme and those that dropped out were similarly not analysed. Two of the outcome measures were developed by the author and it is not clear if these were validated.

The Desouza paper also attracted a '-' score because there was no control group and a high number of people were excluded, ineligible or dropped out which gave a very selective sample. The effectiveness of the programme was also assessed at a fixed point in time which meant that some participants had only just completed the intervention while others took part nine years ago.

The Bewley evaluation had fewer limitations and was based on comparisons of IB recipients in pilot areas and areas selected as control, although each of the areas did not start the intervention at the same time and the authors had some concerns about the use of different sample frames which limited its rating to '+'.

The first two studies provide insufficient evidence to judge the effectiveness of the interventions. Furthermore their applicability to the wider UK population of those on Incapacity Benefit is also limited as the interventions were targeted at the two specific groups. On the other hand, Bewley does provide evidence to indicate that the changes to what happens to potential claimants when the first make an application for IB and the support they are offered to return to work at that points could be effective across Great Britain, if not the UK, at increasing their rate of employment.

4 Cost Effectiveness Findings

4.1 Cost-effectiveness results

No studies were found that satisfied the inclusion criteria for this cost-effectiveness review (see 2.7.2 above). The full papers of 22 potentially relevant primary studies were requested and they have all been received. However, each of these papers have subsequently been excluded. A paper could be excluded for failing to satisfy more than one of the necessary criteria, but for ease of reporting the most obvious excluding criterion is reported below. As a result, no data extraction or quality assessment of any studies, and no synthesis was undertaken.

Table 4.1: Reasons for exclusion of cost-effectiveness studies						
Study	Not UK population	Not a population in receipt of benefit	Not an economic evaluation			
Abasolo et al (2005)	√					
Ballegaard et al (1996)	\checkmark					
Brewer et al (2006)		\checkmark				
Campolieti M. (2001)	~					
CCOHTA (1992)	√					
Cullberg J, et al. (2006)	\checkmark					
Fritz JM, et al (2003)	\checkmark					

Study	Not UK population	Not a population in receipt of benefit	Not an economic evaluation
Gatchel et al (2003)	√		
Gill (1996)		\checkmark	
Hunter SJ, Shaha S, Flint D, Tracy DM. (1998)			\checkmark
Jordan KD et al (1998)			\checkmark
Leon AC, Walkup JT, Portera L. (2002)	✓		
Miller et al (2002)		✓ **	
Monpere et al (2000)	\checkmark		
Pinnington MA, Miller J, Stanley I. (2004)		\checkmark	
Seekamp A, Regel G, Tscherne H. (1996)	✓		
Shaw W et al (2006)	\checkmark		
Thomas C, Morris S. (2003)			\checkmark
Turk DC, Okifuji A.(1998)	\checkmark		
Turk D et al (2001)	✓		
Turk D et al (2004)	\checkmark		
Zeitzer (1991)	\checkmark		
* See Figure2.2 above	e		

** Only one-fifth of population in receipt of benefit and results for this population are not given separately

4.2 Applicability of the evidence to the UK populations in the scope

The absence of evidence here does not permit any comment on the cost effectiveness of interventions to help people back to work or to prepare for work who are resident in the UK and in receipt of incapacity benefit. This absence of economic evaluation is not surprising given the very small number of studies for this population and topic (see Section 3).

5 Discussion

5.1 Effectiveness Findings

The effectiveness review for Q4 identified only three papers which met the inclusion criteria for the review.

5.1.1 Cognitive behavioural therapy

The first study(Winspear, 2007) is a randomised controlled trial of a cognitive behavioural intervention therapy (CBT) and was rated as being of low (ie '-') methodological quality. The intervention consisted of an eight week programme of varying numbers of CBT based sessions (no information on session content is given) being delivered to Incapacity Benefit recipients. Limited information is given about return to work outcomes and it is difficult to identify from the paper whether increases in employability apply to all the intervention group, or only those who completed the intervention. As a result there is insufficient evidence from one study to assess the effectiveness of CBT in improving employment outcomes for incapacity benefit recipients with mild, moderate or sever anxiety or depression and expressing a wish to return to work.

5.1.2 Work focussed interviews plus access to employability support

The second study (Bewley et al, 2007) was a 'non-randomised controlled study', rated as being of medium methodological quality and provides evidence that participation in a Pathways to Work pilot programme resulted in a small, positive increase in the number of those in employment at 18 months (when compared to those who would have been in work at 18 months without the Pathways to Work intervention). The Pathways to Work pilot programme consisted of several elements including:

- a compulsory work focused interview (and up to five compulsory follow up interviews) and voluntary participation in any or all of :
 - o training, and/or
 - o a return to work credit, and/or
 - o in work support
- Additionally, advisers had access to a discretionary fund to make purchases of up to £100 to help people find work.

The study provide sufficient evidence to indicate the intervention is likely to be effective at increasing the chances of people on Incapacity Benefit (IB) being in work 18 months after initially enquiring about accessing IB. The study also found that the employment effects were stronger for women than men and those aged under, rather than over, 50.¹

5.1.3 Rehabilitation programme

The remaining paper (Desouza, 2007) was a case series study rated as being of poor methodological quality. The study examined the impact of an early rehabilitation programme (The Papworth Trust's Early Rehabilitation Programme) comprising of multiple inputs from:

- Rehabilitation co-ordinator
- Case manager
- Job coach
- Vocational advisor
- Information technology assessor
- Vocational psychologist

The programme is open to anyone receiving incapacity benefit and aims to provide participants with employment, medical and life skills over a four to ten month period to enable them to return to work. The study found that 53 of 94 participants were in employment at the point of follow up. The lack of a comparison group means it is not possible to say if this is a better or worse outcome than 'care as usual'. As a result there was insufficient evidence to assess the efficacy of this particular rehabilitation programme.

¹ These findings are supported by a range of other studies about the Pathways programme which are not included in this review as they do not meet the inclusion criteria.

Overall, it should be born in mind that the quality ratings for different types of study design cannot be directly compared and that a weak RCT such as Winspear is a more robust study design than either the before and after or case series research reported by Bewley and Desouza respectively.

5.2 Cost-effectiveness findings

The cost-effectiveness review was unsuccessful in identifying any economic evaluations of interventions to help UK recipients of incapacity benefit to return to work.

5.3 Conclusions

The search for the literature was comprehensive and considerable effort was made to limit bias in the identification, selection, extraction and appraisal of the literature. Overall, this review has identified a paucity of evidence in relation to the effectiveness of interventions and a complete lack of evidence for the cost effectiveness of interventions that met the requirements of the research question and the inclusion and exclusion criteria (eg UK studies based on an RCT or longitudinal methodology).

Two conclusions are apparent. Firstly that the effectiveness and cost effectiveness reviews for research question 4 have identified a need to generate new research in this area as there is so little research which meets the specified inclusion criteria and evaluates the impact of interventions, strategies, programmes and policies to aid those on incapacity benefits to return to work. Second, some form of economic evaluation of interventions that appear to be effective would be helpful (in this case the only example being the Pathways to Work pilot reported on by Bewley et al (2007).¹

This clearly represents a gap in the evidence base, at least in terms of the types of primary studies included in this review. It can be concluded that there is insufficient UK information of this sort on which to base detailed policy and practice, but that this report identifies clear gaps for future research.

In the absence of a sufficient research base, one option is to take into account evidence from other OECD countries. The different contexts and benefit regimes operating in other countries will undoubtedly limit the transferability of any findings (and this was one of the reasons that this review focussed just on the UK). However, given the time it would take to build up a substantial body of evidence, future reviews of this area may benefit from developing a way of at least drawing out the principles operating in other countries to see if any lessons can be learnt for the UK

¹ Reference to a cost benefit analysis is made in the range of literature about Pathways to Work, but no published study was found during searching

References

- Bewley H, Dorsett R, Haile G (2007) The Impact of Pathways to Work, DWP Research Report 435
- Desouza M, Sycamore M, Little S and Kirker S G B, (2007) 'The Papworth early rehabilitation programme: Vocational outcomes', Disability and Rehabilitation, April 2007, 29 (8), p 671-677.
- Drummond MF, Jefferson TD. Guidelines for authors and peer reviewers of economic submissions to the BMJ. British Medical Journal 1996;313:275-83.
- DWP/DH/HSE (2005), *Health, work and well-being: Caring for our future,* HM Government
- Health and Safety Executive (2004), *Managing sickness absence in the public sector: A joint review by the Ministerial Task Force for Health,* Safety and Productivity and the Cabinet Office
- NHS CRD (2001), Undertaking systematic reviews of research on effectiveness, NHS Centre for Reviews and Dissemination, Report No 4, University of York
- Waddell G, Burton AK (2006), *Is work good for your health and well-being?*, Department of Work and Pensions
- Winspear D, 'Using CBT to improve employment outcomes for incapacity benefit customers: Interim report', Journal of Occupational Psychology, Employment and Disability, Vol 9, No 1, Spring 2007, p 41-51.

Appendix 1: Example Search Strategy Used for Research Question 4

MEDLINE primary study search strategy research

new deal adj2 disabled pathways to work return adj2 work adj5 disabilit\$ return adj2 work adj5 incapacity sickness adj3 benefit\$ invalidity adj3 benefit\$ incapacity adj3 benefit\$ disability adj3 benefit\$ sickness adj3 leave invalidity adj3 leave incapacity adj3 leave disability adj3 leave sickness adj3 allowance\$ invalidity adj3 allowance\$ incapacity adj3 allowance\$ disability adj3 allowance\$

sickness adj3 pension\$

invalidity adj3 pension\$

incapacity adj3 pension\$

disability adj3 pension\$

sickness adj3 payment\$

invalidity adj3 payment\$

incapacity adj3 payment\$

disability adj3 payment\$

DISABILITY-INSURED

NDDP

IB adj5 (incapacity or benefit\$)

access to work

or/1-28

Appendix 2: Sifting Criteria Used

The following pages constitute the NICE Absence Sift Criteria for Question 4.

NICE Absence Sift Criteria: Effectiveness primary studies Q4

The ultimate aim of Question 4 of the review is to provide guidance on what interventions, programmes, policies or strategies are effective and cost effective in helping those in receipt of incapacity benefit to return to full or part time work?

For all titles and abstracts answer ALL questions UNLESS excluded at any stage

	QUESTION	ANS	QUALIFIER	ACTION
1	Is this a primary study evaluation of an intervention, policy, strategy or programme which focuses on helping	No	Review/ Systematic review or meta analysis	Code as rev
			Economic data	Code as Econ
	benefit (or another (or		Other	Exclude
	previous) sickness/ill health related benefit			
	such as invalidity benefit)	Yes	Doc type dissertation	Exclude
	return to work?		Intervention – deals solely with the provision of pharmacological treatment	Exclude
			Intervention – deals solely with effectiveness of IB system, private health insurance schemes or statutory sick pay	Exclude
			Intervention – delivered in a non- primary care setting without primary care or employer involvement in its planning, design, delivery, management or funding	Exclude
			Other	Go to Q2
		Unclear		Code as unclear & go to Q2
2	Is the study based in the UK?	No		Exclude
		Yes		Go to Q3
		Unclear		Code as unclear &

-

	QUESTION	ANS	QUALIFIER	ACTION		
				Go to Q3		
3	Does the study measure return to work related	No		Exclude		
	outcomes?					
		Yes	Return to full or part time employment	Code as R4 & go to Q4		
			Some other work related outcome	Code as unclear		
		Unclear				
4	Is the study longitudinal?	No		Exclude		
		Yes		Include		
		Unclear		GFP		
	Enter decision – if any 'unclear' code as GFP					
Q4	Book chapter?	Yes		Тад		

NICE Absence Sift Criteria:

Economics (primary studies and reviews) Q4

The ultimate aim of the economics reviews is to provide guidance on the most cost effective actions to manage sickness absence and support return to work or to help those in receipt of incapacity benefit to return to full or part time work

	QUESTION	ANS	QUALIFIER	ACT'N
Q1	Is the study a comparative economic evaluation of an	Yes	Population – if focuses solely on children (16 & under)	Exclude
	intervention, policy, strategy or programme which focuses on helping recipients of incapacity benefit (or another (or previous) sickness/ill health related benefit such as invalidity benefit) return to work? <i>ie study type is "cost- effectiveness", "cost consequences", "cost benefit", "cost-utility", "cost-minimisation" or reports outcomes cost per x gained or cost per x avoided or net monetary benefit (cost benefit only). Code as unclear anything with costs and consequences reported separately but doesn't call it any of the above. <i>ie. not costs of an illness</i></i>		Intervention – delivered in a non- primary care setting without primary care or employer involvement in its planning, design, delivery, management or funding delivered solely in a developing country	Exclude
			Absence - not due to sickness (eg maternity leave)	Exclude
			Topic – if focuses solely on pharmacological treatment	Exclude
			Topic – if focus is prevention of 1 st instance of sickness absence	Exclude
			Topic – if focuses solely on effectiveness of IB system, private health insurance schemes or statutory sick pay	Exclude
			Abstracts describes comparative economic evaluation of an intervention, policy etc	Go to Q2
		Unclear	Abstracts unclear if comparative economic evaluation of an intervention, policy etc	Code U/C & go to Q2
			Abstracts includes work related costs & consequences but not comparative economic evaluation of	Code - Modellin g

For all titles and abstracts answer ALL questions UNLESS coded as exclude at any stage

-

	QUESTION	ANS	QUALIFIER	ACT'N
			intervention, policy etc	
		No		Exclude
Q2	Does the study contain work related outcome	No		Exclude
	measures?			
		Yes	Return to full or part time employment	Go to Q3
			Some other work related outcome	Code Oth
		Unclear		Code U/C
Q3	Is the study based in UK	No	If relevant to R1, R2, R3, include there, if not, exclude	Exclude
		Yes		Code as Review 4
		Unclear		Code U/C
	Enter decision in	final colui	mn – if any 'unclear' code as GFP	
		Tag if k	book chapter	

NICE Absence Sift Criteria: Effectiveness reviews Q 1 to 3 and Q4

The ultimate aim of the review is to provide guidance on the most effective actions to manage sickness absence and support return to work or to help those in receipt of incapacity benefit to return to full or part time work

For all titles and abstracts answer ALL questions UNLESS coded as exclude at any stage

	QUESTION	ANS	QUALIFIER	ACTION
Q1	Is the article a review of interventions, policies, strategies or	Yes	Population – if focuses solely on children (16 & under)	Exclude
	programmes aimed at reducing sickness absence or aiding return to work?		Setting – if related to reducing sickness absence and delivered in non workplace or non primary care setting (check IB relevant)	Exclude
			Setting - delivered solely in a developing country	Exclude
			Absence - not due to sickness (eg maternity leave)	Exclude
			Intervention – if solely health promotion or prevention of 1st instance	Exclude
			Intervention – if focuses solely on pharmacological treatment	Exclude
			Intervention – deals solely with effectiveness of IB system, private health insurance schemes or statutory sick pay	Exclude
			Intervention, policy etc – other	Go to Q2
		Unclear	Intervention, policy etc – unclear	unclear & go to Q2
		No		Exclude
Q2	Is it a review of	Yes		Go to

	QUESTION	ANS	QUALIFIER	ACTION
	longitudinal studies?			Q3
		No		Exclude
		Unclear		unclear & go to Q3
Q3	Does the review report	No		Exclude
	measures?			
		Yes	Reduce the numbers of employees moving from short to long term sickness absence?	code Review 1
			Aid the return to work of employees after long term sickness absence?	code Review 2
			Reduce the re-occurrence of an employees long term sickness absence?	code Review 3
			Assist UK incapacity benefit recipients in returning to work	Go to Q4
			Other work-related outcome	Code Other
		Unclear		Code unclear
Q4	Does the review contain data from (a) UK based	No		Exclude
	study/ies			
		Yes		Include
		Unclear		Code unclear
	TAG		Book chapter	Tag
			Economics data	Tag
			Primary study	Tag
	Enter decision i	in final colu	umn – if any 'unclear' code as GFP	

Appendix 3: Full Paper Screening Checklists

Full paper screening TRIAGE Form: Q4 Effectiveness (primary study)	Reference Manager ID No:		
Paper checked by: Date of check:			
Population : Is the study an evaluation of an intervention/policy/strategy/programme which aims to help adults (over 16) who are unemployed and in receipt of incapacity benefit (or a previous form of incapacity benefit or other similar benefit) return to work (paid and unpaid)?	Yes	No	unclear
Is the study set in the UK?	Yes	No	unclear
Intervention : Is the intervention/policy/strategy/programme being delivered in a primary-care setting and/or workplace setting and/or planned, designed, delivered, managed or funded in collaboration with primary care providers and/or employers? These interventions etc. can be delivered by a number of providers (such as voluntary, private, statutory sector) and/or in various settings not just workplace or primary care setting (such as job centres, community centres) as long as they are fully or co-planned, designed, delivered, managed and/or funded in collaboration with employers and primary care service). Interventions can include mixed component studies - eg treatment and public health	Yes	No	unclear
Outcome: Is one of the following outcomes being measured:			
Return to work (paid/unpaid)1 Sustained return to work (paid/unpaid) No effect on return to work, job-related activity or any other work related outcome	Yes Yes Yes	No No No	unclear unclear unclear
If yes to Q4, are any other work related outcomes measured (ie uptake of or increased job seeking; increase in work experience and vocational training and increase in skills/knowledge for work/unpaid work or alternative career/work) Yes No Unclear			
Study Design: Is the study longitudinal in design (ie at least one measurement after baseline)If yes to Q6, What is the study design?: RCT	Yes	No	Unclear

Controlled before and after			
Cohort			
Case control			
Before and after			
Interrupted Time Series			
Other (please specify)			
IF ANY 1-5 = 'NO', CHECK Q8 THEN EXCLUDE			
If 1-5 all Yes , does the study meet any of the following exclusion criteria:			
Study – pre 1990	Yes	No	unclear
Study not English language	Yes	No	unclear
Deals solely with the provision of treatment for existing conditions (including pharmacological or therapeutic interventions)	Yes	No	unclear
Deals solely with the effectiveness of the incapacity benefit system, private health insurance schemes or statutory or occupational sick pay.			
Deals solely with preventing ill-health retirement (ie where recipient has no intention of returning to work)	Yes	No	unclear
IF ANY Q6 = 'YES', CHECK Q8 THEN EXCLUDE	Yes	No	unclear
IF ANY Q1-6 = 'UNCLEAR', CHECK Q8 THEN REFER FOR SECOND OPINION			
If the study is a combined intervention, can the public health data be disaggregated?	Yes	No	Unclear
Does the study contain economic/cost data or effectiveness data relevant to the other research questions* OR data relevant for the economic	Q1		
modelling for this research question (which does not have to be UK)	Q2		
	Q3		
IF Q7 = 'NO', TAG AS MIXED, CHECK Q8 THEN EXCLUDE	Mode	lling	
IF ANY Q1-7 = 'UNCLEAR', CHECK Q8 THEN REFER FOR SECOND OPINION		Ŭ	
Put forward for QUALITY ASSSESSMENT and DATA EXTRACTION (studies to be grouped by study design type)			

(*Q1 Preventing/reduce employees moving from short to long term sickness; Q2 Help employees return to work from LTSA; Q3 prevent the re-occurrence of LTSA)

Full paper screening TRIAGE Form: Q4 Economics (primary study)	Refe	Reference Manager ID No:	
Paper checked by: Date of check:			
Population: Is the study an evaluation of an intervention/policy/strategy/programme which aims to help adults (over 16) who are unemployed at receipt of incapacity benefit (or a previous form of incapacity benefit or other similar benefit) return to work (paid and unpaid)?	nd in Yes	No	unclear
Is the study set in the UK?	Yes	No	unclear
Intervention: Is the intervention/policy/strategy/programme being delivered in a primary-care setting and/or workplace setting and/or planned, designed, delivered, managed or funded in collaboration with primary care providers and/or employers These interventions etc. can be delivered a number of providers (such as voluntary, private, statutory sector) and/or in various settings not just workplace or primary care setting (such as centres, community centres) as long as they are fully or co-planned, designed, delivered, managed and/or funded in collaboration with employee and primary care service)?Interventions can include mixed component studies (eg treatment and public health studies)	d by job rs	Νο	unclear
Outcome: Is one of the following outcomes being measured: Return to work (paid/unpaid) Sustained return to work (paid/unpaid) No effect on return to work, job-related activity or any other work related outcome	Yes Yes Yes	No No No	unclear unclear Unclear
If yes to Q4, are any other work related outcomes measured (ie uptake of or increased job seeking; increase in work experience and vocat training and increase in skills/knowledge for work/unpaid work or alternative career/work) Yes No unclear	ional		
Study design: Is the study an economic evaluation (ie an RCT or longitudinal study with at least one follow up measure after baseline) with cost effectiveness, cost benefit, cost utility, cost minimization or net monetary (cost) benefit data?	t Yes	No	unclear
If yes to Q5, What is the study design? Cost benefit (CBA) Cost effectiveness Cost utility Other			
IF ANY 1-5 = 'NO', CHECK Q8 THEN EXCLUDE			
Study – pre 1990 Study not English language Deals solely with the provision of treatment for existing conditions (including pharmacological or therapeutic interventions) Deals solely with the effectiveness of the incapacity benefit system, private health insurance schemes or statutory or occupational sick Deals solely with preventing ill-health retirement (ie where recipient has no intention of returning to work)	Yes Yes Yes Yes Yes Yes	No No No No	unclear unclear unclear unclear unclear
IF ANY Q6 = 'YES', CHECK Q8 THEN EXCLUDE			
If the study is a combined intervention, can the public health data be disaggregated? Does the study contain economic/cost data or effectiveness data relevant to the other research questions* OR data relevant for the economic modelling for this research question (which does not have to be UK)	Yes No Unclear Q1 Q2 Q3		
---	----------------------------------		
IF Q7 = 'NO', TAG AS MIXED, THEN EXCLUDE IF ANY Q1-7 = 'UNCLEAR', REFER FOR SECOND OPINION	Modelling		
Put forward for QUALITY ASSSESSMENT and DATA EXTRACTION (studies to be grouped by study design type)			

* Q1 – Preventing/reduce employees moving from short to long term sickness; Q2 Help employees return to work from LTSA; Q3 prevent the re-occurrence of LTSA

Full paper screening TRIAGE Form: Q4 Reviews	Ref N	lan ID I	No:
Paper checked by: Date of check:			
Population : Does the review partly or wholly cover evaluations of an intervention/policy/strategy/programme which aims to help adults (over 16) who are unemployed and in receipt of incapacity benefit (or a previous form of incapacity benefit or other similar benefit) return to work (paid and unpaid)?	Yes	No	unclear
Is at least one of the studies in the review set in the UK?	Yes	No	unclear
Intervention : Is at least one of the studies in the review an evaluation of an intervention/policy/strategy/programme being delivered in a primary-care setting and/or workplace setting and/or planned, designed, delivered, managed or funded in collaboration with primary care providers and/or employers These interventions etc. can be delivered by a number of providers (such as voluntary, private, statutory sector) and/or in various settings not just workplace or primary care setting (such as job centres, community centres) as long as they are fully or co-planned, designed, delivered, managed and/or funded in collaboration with employers and primary care service)?	Yes	No	unclear
Outcome: Is one of the following outcomes being measured: Return to work (paid/unpaid) Sustained return to work (paid/unpaid) No effect on return to work, job-related activity or any other work related outcome If yes to Q4, are any other work related outcomes measured (ie uptake of or increased job seeking; increase in work experience and vocational training and increase in skills/knowledge for work/unpaid work or alternative career/work)	Yes Yes Yes	No No No	unclear unclear unclear
Study design: Does the review include at least one RCT or longitudinal study (eg. the same data collected from at least 2 different points in time)? If yes to Q6 does it contain effectiveness studies or economic evaluations with cost effectiveness, cost benefit, cost utility, cost consequences, cost	Yes	No	unknown
IF ANY 1-5 = 'NO', CHECK Q8 THEN EXCLUDE	Yes	No	unknown
If 1-5 all Yes , does the review meet any of the following exclusion criteria: Review – pre 1990 Review not English language Deals solely with the provision of treatment for existing conditions (including pharmacological or therapeutic interventions) Deals solely with the effectiveness of the incapacity benefit system, private health insurance schemes or statutory or occupational sick pay. Deals solely with preventing ill-health retirement (ie where recipient has no intention of returning to work) IF ANY Q7 = 'YES', CHECK Q8 THEN EXCLUDE	Yes Yes Yes Yes Yes	No No No No	unclear unclear unclear unclear unclear
*Does the review contain economic/cost data or effectiveness data relevant to the other research questions OR data which may be useful for the economic modelling (specify)	Yes	No	unclear

If any Q1-8 ='UNCLEAR', REFER FOR SECOND OPINION	
Check review references against Ref Man and if not in Ref Man file order for abstract appraisal	
* Q1 – Preventing/reduce employees moving from short to long term sickness; Q2 Help employees return to work from LTSA; Q3 prevent the re-oc	currence of LTSA

Appendix 4: Excluded Primary Studies by Reason for Exclusion

Effectiveness studies: Excluded studies by reason

Non intervention, policy, programme or strategy

- Anema, J. R., Van Der Giezen, A. M., Buijs, P. C., & van Mechelen, W. 2002,
 "Ineffective disability management by doctors is an obstacle for return-to-work: a cohort study on low back pain patients sicklisted for 3-4 months", *Occupational* & Environmental Medicine, vol. 59, no. 11, pp. 729-733.
- Ashworth, K., Hartfree, Y., & Stephenson, A. 2001, "Well enough to work?", *X*, vol. DSS-RR-145, p. 130.
- Bean, S. 2005, "DWP is helping incapacity claimants get back to work", *Occupational Health*, vol. 57, no. 11, p. 11.
- Beinart, S. 1997, "The Access to Work Programme. Further analysis of data from the 1995 surveys of ATW recipients and their employers", _, vol. SCPR-R-1607, p. 61.
- Bennie, M. N. 1789, "Incapacity Benefit Reform Pilot: Condition Management Programme Evaluation Complete,".
- Blonk, R. W. B., Brenninkmeijer, V., Lagerveld, S. E., & Houtman, I. L. D. 2006,
 "Return to work: A comparison of two cognitive behavioural interventions in cases of work-related psychological complaints among the self-employed", *Work* & Stress, vol. 20, no. 2, pp. 129-144.
- Brattberg, G. 2006, "Internet-based rehabilitation for individuals with chronic pain and burnout: a randomized trial", *International Journal of Rehabilitation Research*, vol. 29, no. 3, pp. 221-227.

- Calkins, J. 2001, "News from the CDMSC: Certification of Disability Management Specialists Commission. Return-to-work programs need communications, cooperation", *Care Management*, vol. 7, no. 2, p. 8.
- Carosella, A. M., Lackner, J. M., & Feuerstein, M. 1994, "Factors associated with early discharge from a multidisciplinary work rehabilitation program for chronic low back pain", *Pain*, vol. 57, no. 1, pp. 69-76.
- Collado-Ramos, P., Richi, P., Crespo, M., Revenga, M., Bachiller, J., Candelas, G., Blanco, M., & Jover, J. A. 2004, "Painful upper limb disorders and disability among working population: Effectiveness of a specific return to work program", *Annals of the Rheumatic Diseases*, vol. 63, p. 507.
- Damiani, G., Federico, B., Pinnarelli, L., & Ricciardi, G. 2004, "Do occupational stress management programmes affect absenteeism rates?", *Occupational Medicine-Oxford*, vol. 54, no. 1, pp. 58-59.
- Davis A, Pound E, Stafford B (2006), *New Deal for Disabled People Extensions: examining the role and operation of new Job Brokers*, DWP Research Report No. 384
- De Souza, L. (2007). A single blind randomised trial to evaluate physiotherapy for back pain patients. *Department of Health.*
- Dent, E. 2006, "Walk the talk", Health Service Journal, vol. 116, no. 6017, pp. 22-24.
- Dewson S (2005), Evaluation of the Working Neighbourhoods Pilot: Year One, DWP Research Report No. 297
- Dewson S, Ritchie H, Meager N (2005), New Deal for Disabled People: Survey of Employers, DWP Research Report No. 301
- Diffendal, V. & White, W. 2000, "Trends. Integrated disability management: will it gain momentum?", *AWHP'S Worksite Health*, vol. 7(1), no. 12-3, p. 19.
- Dozois, D. J. A., Dobson, K. S., Wong, M., Hughes, D., & Long, A. 1995, "Factors associated with rehabilitation outcome in patients with low back pain (LBP): Prediction of employment outcome at 9-month follow-up", *Rehabilitation Psychology*, vol. 40, no. 4, pp. 243-259.
- Erens, B. & Ghate, D. 1993, "Invalidity benefit. A longitudinal survey of new recipients", _, vol. DSS-RR-20, p. 137.
- Fritz, J. M., Delitto, A., & Erhard, R. E. 2003, "Comparison of classification-based physical therapy with therapy based on clinical practice guidelines for patients with acute low back pain: a randomized clinical trial", *Spine*, vol. 28, no. 13, pp. 1363-1371.
- Gilbert, P. P. 1789, "Evaluation of the Condition Management Programme Ongoing,".

- Goss, N. 1996, "Access to work: the end of the success story?", *Rehab Network*, vol. 41, pp. 18-19.
- Hazard, R. G., Haugh, L. D., Reid, S., McFarlane, G., & MacDonald, L. 1997, "Early physician notification of patient disability risk and clinical guidelines after low back injury A randomized, controlled trial", *Spine*, vol. 22, no. 24, pp. 2951-2958.
- Highley, H. 2003, "Leadership", *Primary Care Special Report* 2003; (7): (17 November 2003) p. 8.
- Holmes, P. & Lynch, M. 1990, "An analysis of invalidity benefit claim durations for new male claimants in 1977/1978 and 1982/1983", *Journal of Health Economics*, vol. 9, no. 1, pp. 71-83.
- Huibers, M. J. H., Beurskens, A. J. H. M., Van Schayck, C. P., Bazelmans, E., Metsemakers, J. F. M., Knottnerus, J. A., & Bleijenberg, G. 2004, "Efficacy of cognitive-behavioural therapy by general practitioners for unexplained fatigue among employees: Randomised controlled trial", *British Journal of Psychiatry*, vol. 184, pp. 240-246.
- Jellema, P., van der Windt, D., van der Horst, H. E., Twisk, J. W. R., Stalman, W. A. B., & Bouter, L. M. 2005, "Should treatment of (sub)acute low back pain be aimed at psychosocial prognostic factors? Cluster randomised clinical trial in general practice", *British Medical Journal*, vol. 331, no. 7508, pp. 84-87.
- Karjalainen, K., Malmivaara, A., Mutanen, P., Roine, R., Hurri, H., & Pohjolainen, T. 2004, "Mini-intervention for subacute low back pain - Two-year follow-up and modifiers of effectiveness", *Spine*, vol. 29, no. 10, pp. 1069-1076.
- Kazimirski A, Adelman L, Arch J, Keenan L, Legge K, Shaw A, Stafford B, Taylor R, Tipping S (2005), New Deal for Disabled People Evaluation: Registrants Survey Merged Cohorts (Cohorts one and two, Waves one and two), DWP Research Report No. 260
- Killoughery, M. 1999, "Disability and incapacity benefits. The role that doctors play", *Psychiatric Bulletin*, vol. 23, no. 5, pp. 260-263.
- Knight, E. 1992, "Claiming disability benefits.[erratum appears in Health Visit 1992 Nov;65(11):415]", *Health Visitor*, vol. 65, no. 9, p. 323.
- la-Posta, C. & Drummond, P. D. 2006, "Cognitive behavioural therapy increases reemployment of job seeking worker's compensation clients", *Journal of Occupational Rehabilitation*, vol. 16, no. 2, pp. 223-230.
- Legge K, Magadi M, Phung V G, Stafford B, Hales J, Hayllar O, Nevill C, Wood M (2006), *New Deal for Disabled People: Survey Registrants – report of Cohort 3*, DWP Research Report No. 369

- Leone, S. S., Huibers, M. J. H., Kant, I., Van Schayck, C. P., Bleijenberg, G., & ndre Knottnerus, J. 2006, "Long-term predictors of outcome in fatigued employees on sick leave: a 4-year follow-up study", *Psychological Medicine*, vol. 36, no. 9, pp. 1293-1300.
- Lindstrom, I., Ohlund, C., Eek, C., Wallin, L., Peterson, L. E., Fordyce, W. E., & Nachemson, A. L. 1992, "The effect of graded activity on patients with subacute low back pain: a randomized prospective clinical study with an operantconditioning behavioral approach", *Physical Therapy*, vol. 72, no. 4, pp. 279-290.
- Lonsdale, S., Lessof, C., & Ferris, G. 1993, "Invalidity benefit. A survey of recipients", vol. DSS-RR-19, p. 70.
- Lotters, F., Hogg-Johnson, S., & Burdorf, A. 2005, "Health status, its perceptions, and effect on return to work and recurrent sick leave", *Spine*, vol. 30, no. 9, pp. 1086-1092.
- Loumidis J, Stafford B, Youngs R, Green A, Arthur S, Legard R, Lessof C, Lewis J, Walker R, Corden A, Thornton P, Sainsbury R (2001), *Evaluation of the New Deal fro Disabled People Personal Adviser Service Pilots*, DWP Research Report No. 144
- Lynn, J. 2003, "Get with the Program!", Entrepreneur, vol. 31, no. 7, p. 75.
- Main, P. C. 1 A.D., "Healthy workers healthy companies: Moving pain management to the source Complete,".
- Maisiak, R. S., White, M. B., Overman, L. B., Paolone, F. J., & Fine, P. 2000, "Health status (SF-36) barriers to return to work for persons with musculoskeletal disability", *Arthritis and Rheumatism*, vol. 43, no. 9, p. S285.
- Mangan, D. 2001, "Keep that insurer from blocking your disability claim", *Medical Economics*, vol. 78, no. 15, pp. 57-58.
- Massie, B. 2000, "Getting disabled people to work", _, vol. EPI-ER-14/7, p. 12.
- Mattaliano, R. 2006, "Educating smaller employers on the benefits of expanding disability management", *Contemporary Rehab*, vol. 62, no. 7, p. 10.
- Moncrieff, J. 1999, "The defeat depression campaign and trends in sickness and invalidity benefits for depressive illness", *Journal of Mental Health*, vol. 8, no. 2, pp. 195-202.
- Morrison, D. J. "Depression and long term work incapacity in Scotland: the role of the GP Ongoing,".
- Norwood, M. H. "Factors affecting self-declared long term sickness and disability in a South Yorks town: a qualitative narrative approach Complete,".

- Ockander, M., Timpka, T., & Nyce, J. M. 2005, "How to avoid long-term sickness absence: the advice from women with personal experience", *Family Practice*, vol. 22, no. 4, pp. 394-398.
- Orr L L, Bell S H, Lam K (2007), *Long-term impacts of the New Deal for Disabled People*, DWP Research Report No. 432
- Pires C, Kazimirski A, Shaw A, Sainsbury R, Meah A (2006), *New Deal for Disabled People Evaluation: Survey Eligible Population, Wave Three,* DWP Research Report No. 324
- Redway H (2001), New Deal for Disabled People: Using administrative data to access the impact on exits from benefit, DWP In-house Report No. 81
- Ross, M. J. 1 A.D., "What Factors Influence the Practice of Occupational Therapy in Vocational Rehabilitation and how can Practice be Understood from an Occupation - Focused Perspective? Ongoing,".
- Stikeleather, J. 2004, "An older worker's decision to "push or protect self" following a work-related injury", WORK: A Journal of Prevention, Assessment & Rehabilitation, vol. 22, no. 2, pp. 139-144.
- Strand, L. I., Ljunggren, A. E., Haldorsen, E. M., & Espehaug, B. 2001, "The impact of physical function and pain on work status at 1-year follow-up in patients with back pain", *Spine*, vol. 26, no. 7, pp. 800-808.
- Sullivan, M. J., Adams, H., Thibault, P., Corbiere, M., & Stanish, W. D. 2006, "Initial depression severity and the trajectory of recovery following cognitive-behavioral intervention for work disability", *Journal of Occupational Rehabilitation*, vol. 16, no. 1, pp. 63-74.
- Tate, D. G. 1992, "Factors influencing injured employees' return to work", *Journal of Applied Rehabilitation Counselling*, vol. 23, no. 2, pp. 17-20.
- Taylor, A. 2003, "Access to Work blocked as disabled people remain unaware of initiative", *Community Care* no. 1486, pp. 16-17.
- Taylor, D. S. "Improving the Effectiveness of Referrals to the Occupational Health Department under the Southampton University Hospitals NHS Trust Sickness Absence Policy Complete,".
- Thornton, P. 2002, "Disability", Research Matters, vol. Oct 2002, p. Apr.
- van Duijn, M., Lotters, F., & Burdorf, A. 2004, "Interrelationships between pain, disability, general health, and quality of life and associations with work-related and individual factors - A study among workers on sickness absence for 2 to 6 weeks for musculoskeletal complaints", *Spine*, vol. 29, no. 19, pp. 2178-2183.

- Vendrig, A. A. 1999, "Prognostic factors and treatment-related changes associated with return to work in the multimodal treatment of chronic back pain", *Journal of Behavioural Medicine*, vol. 22, no. 3, pp. 217-232.
- Verbeek, J. H., van der Weide, W. E., & van Dijk, F. J. 2002, "Early occupational health management of patients with back pain: a randomized controlled trial", *Spine*, vol. 27, no. 17, pp. 1844-1851.
- Waddell, G. 2006, "Preventing incapacity in people with musculoskeletal disorders", *British Medical Bulletin*, vol. 77-78, pp. 55-69.
- Walker, J. M. 2003, "Disabler: a game occupational health nurses cannot afford to play", *AAOHN Journal*, vol. 51, no. 10, pp. 421-424.
- Ward, S. 2005, "Closing the inequalities gap in Rotherham", *Primary Care Report*, vol. no 8, p. Sep.
- Watson, P. J. & Main, C. J. 2004, "Influence of benefit type on presenting characteristics and outcome from an occupationally orientated rehabilitation programme for unemployed people with chronic low back pain", *Physiotherapy*, vol. 90, no. 1, pp. 4-11.
- Wigham, R. 2005, "Government to launch health drive to get sick back to work", *Employers Law* p. 4.
- Williams, N. 2005, "Fitness for work I: advice and certification", *Practice Nursing*, vol. 16, no. 2, pp. 86-88.
- Zarb, G., Jackson, N., & Taylor, P. 1996, "Helping disabled workers. Disability Working Allowance and supported employment", _, vol. DSS-RR-57, p. 122.

Non UK

- 2001, "Keeping employees at work", Benefits Canada, vol. 25, no. 3, p. 27.
- Angelo, J. 1993, "A model for helping persons with physical disabilities return to work", *Occup Ther Pract*, vol. 4, no. 3, pp. 67-73.
- Arbour, K. P., Latimer, A. E., Jung, M. E., & Ginis, K. A. M. 2004, "Moving beyond the stigma: Self-presentational benefits of exercise in individuals with a physical disability", *Journal of Sport & Exercise Psychology*, vol. 26, p. S27.
- Arnold, R. 1998, "Employment and disability", *Psychiatric Services*, vol. 49, no. 10, p. 1361.
- Aure, O. F., Nilsen, J. H., & Vasseljen, O. 2003, "Manual therapy and exercise therapy in patients with chronic low back pain - A randomized, controlled trial with 1year follow-up", *Spine*, vol. 28, no. 6, pp. 525-531.

- Berglind, H. & Gerner, U. 2002, "Motivation and return to work among the long-term sicklisted: an action theory perspective", *Disability and Rehabilitation*, vol. 24, no. 14, pp. 719-726.
- Beutel, M. E., Zwerenz, R., Bleichner, F., Vorndran, A., Gustson, D., & Knickenberg, R.
 J. 2005, "Vocational training integrated into inpatient psychosomatic rehabilitation--short and long-term results from a controlled study", *Disability & Rehabilitation*, vol. 27, no. 15, pp. 891-900.
- Blumenthal, S. M. 2006, "Inside case management. How to integrate medical management and vocational rehabilitation in disability management programs... reprinted from Inside Case Management, Vol. 5/No. 4 (July 1998)", *Lippincott's Case Management*, vol. 11, no. 3, pp. 183-185.
- Bricout, J. C. 2004, "Using telework to enhance return to work outcomes for individuals with spinal cord injuries", *Neurorehabilitation*, vol. 19, no. 2, pp. 147-159.
- Brucker, D. L. 2004, "Informing the development of employment programs for persons with disabilities: a case study analysis of the definitions, uses and implications of the idea of suitable employment", *Journal of Vocational Rehabilitation*, vol. 21, no. 3, pp. 137-147.
- Cott, A., Anchel, H., Goldberg, W. M., Fabich, M., & Parkinson, W. 1990, "Noninstitutional treatment of chronic pain by field management: an outcome study with comparison group", *Pain*, vol. 40, no. 2, pp. 183-194.
- Curtis, J. & Scott, L. R. 2004, "Integrating disability management into strategic plans: creating healthy organizations", *AAOHN Journal*, vol. 52, no. 7, pp. 298-301.
- De Rijk, A. & Meershoek, A. 2006, "Responsive evaluation of an intervention to promote reintegration to work among people on disability pension", *European Journal of Public Health*, vol. 16, p. 58.
- Dewa, C. S., Hoch, J. S., & Goering, A. P. 2004, "Getting men and women receiving depression-related short-term disability benefits back to work: Where do we begin?", *Economics Of Gender And Mental Illness*, vol. 15, pp. 155-176.
- Difabio, R. P., Mackey, G., & Holte, J. B. 1996, "Physical therapy outcomes for patients receiving workers' compensation following treatment for herniated lumbar disc and mechanical low back pain syndrome", *Journal of Orthopaedic & Sports Physical Therapy*, vol. 23, no. 3, pp. 180-187.
- Doliska, C., Zietkowski, Z., & Bodzenta-Lukaszyk, A. 2005, "Knowledge of risk factors and guidelines for the management of asthma. The educational role of the nurse", *Roczniki Akademii Medycznej W Bialymstoku*, vol. 50 Suppl 1, pp. 177-180.

- Feuerstein, M., Callanharris, S., Hickey, P., Dyer, D., Armbruster, W., & Carosella, A. M. 1993, "Multidisciplinary Rehabilitation of Chronic Work-Related Upper Extremity Disorders Long-Term Effects", *Journal of Occupational and Environmental Medicine*, vol. 35, no. 4, pp. 396-403.
- Guleserian, B. 2007, "A cognitive-behavioral approach improves case management outcomes", *Professional Case Management*, vol. 12, no. 1, pp. 55-59.
- Hakala, P., Karppi, S. L., Marniemi, J., Rastas, M., & Virta, L. 2004, "Perceived working ability and sickness allowances in obese subjects before and after a rehabilitation course", *International Journal of Obesity*, vol. 28, p. S118.
- Jordan, K. D., Mayer, T. G., & Gatchel, R. J. 1998, "Should extended disability be an exclusion criterion for tertiary rehabilitation? Socioeconomic outcomes of early versus late functional restoration in compensation spinal disorders", *Spine*, vol. 23, no. 19, pp. 2110-2116.
- Kennedy, J. & Olney, M. F. 2006, "Factors associated with workforce participation among SSDI beneficiaries", *Journal of Rehabilitation*, vol. 72, no. 4, pp. 24-30.
- Koopman, F. S., Edelaar, M., Slikker, R., Reynders, K., van der Woude, L. H. V., & Hoozemans, M. J. M. 2004, "Effectiveness of a multidisciplinary occupational training program for chronic low back pain", *American Journal of Physical Medicine & Rehabilitation*, vol. 83, no. 2, pp. 94-103.
- Leavitt, R. & Spear, S. 2002, "The power of work: occupational therapy and thorough, standardized assessment can help people with mental health disabilities return to work and, thus, to their communities", *Rehab Management: The Interdisciplinary Journal of Rehabilitation*, vol. 15, no. 5, p. 46.
- Linton, S. J., Boersma, K., Jansson, M., Svard, L., & Botvalde, M. 2005, "The effects of cognitive-behavioral and physical therapy preventive interventions on painrelated sick leave: a randomized controlled trial", *Clinical Journal of Pain*, vol. 21, no. 2, pp. 109-119.
- Lipow, V. A. 1997, "Aging workers: disability management strategies", *Rehab Management*, vol. 10, no. 1, pp. 32-34.
- Martin, M. 2001, "Early Intervention Can Speed Employees' Return to Work", *Occupational Hazards*, vol. 63, no. 7, p. 19.
- Mashburn, K. & Mitchell, K. 2001, "The role of the case manager in an impairmentbased return-to-work process", *Case Manager*, vol. 12, no. 1, pp. 58-61.
- McCollom, P. & Vierling, L. E. 1999, "Case management: a beginning, an end", *Topics in Spinal Cord Injury Rehabilitation*, vol. 4, no. 4, pp. 10-15.
- Nordeman, L., Nilsson, B., Moller, M., & Gunnarsson, R. 2006, "Early access to physical therapy treatment for subacute low back pain in primary health care: a

prospective randomized clinical trial", *Clinical Journal of Pain*, vol. 22, no. 6, pp. 505-511.

- Norrefalk, J. R., Svensson, O., Ekholm, J., & Borg, K. 2005, "Can the back-to-work rate of patients with long-term non-malignant pain be predicted?", *International Journal of Rehabilitation Research*, vol. 28, no. 1, pp. 9-16.
- Park, C. A. 2002, "Right where you are: on-site intervention methods for industrial rehab can save time and money", *Rehab Management: The Interdisciplinary Journal of Rehabilitation*, vol. 15, no. 9, p. 50.
- Piha, K., Harkonmaki, K., Rahkonen, O., & Lahelma, E. 2006, "Pathways between socioeconomic status, long sickness absence, and disability pension", *European Journal of Public Health*, vol. 16, p. 37.
- Reed, P. & Koral, A. M. 2002, "Keep FMLA claims in check: employers can reduce their potential exposure when coordinating FMLA and worker's compensation", *Occupational Health & Safety*, vol. 71(7), no. 70, pp. 72, 74.
- Roberts-Yates, C. 2003, "The concerns and issues of injured workers in relation to claims/injury management and rehabilitation: the need for new operational frameworks", *Disability and Rehabilitation*, vol. 25, no. 16, pp. 898-907.
- Shrey, D. E., Bangs, S. A., Mark, L. S., & Hursh, N. C. 1991, "Returning social security beneficiaries to the work force: A proactive disability management model", *Rehabilitation Counseling Bulletin*, vol. 34, no. 3, pp. 257-273.
- Tortarolo, J. S. & Polakoff, P. L. 1995, "The future of disability management is ... integration", *Benefits Quarterly*, vol. 11, no. 3, pp. 49-55.
- Tufescu, T. V. & Buckley, R. 2001, "Age, gender, work capability, and worker's compensation in patients with displaced intraarticular calcaneal fractures", *Journal of Orthopaedic Trauma*, vol. 15, no. 4, pp. 275-279.
- Von Korff, M., Balderson, B. H. K., Saunders, K., Miglioretti, D. L., Lin, E. H. B., Berry, S., Moore, J. E., & Turner, J. A. 2005, "A trial of an activating intervention for chronic back pain in primary care and physical therapy settings", *Pain*, vol. 113, no. 3, pp. 323-330.
- Wade, E. L. 2002, "Computer applications. Linked by pain: injured and ill workers can find many kinds of help online", *Occupational Health & Safety*, vol. 71(12), no. 20, p. 22.

Population

- Bernkley, T., Henriksen, M., Jahnsen, J., Vatn, M., & Mount, B. 2006, "Sick leave, unemployment, disability and HRQOL in patients with inflammatory bowel disease", *Gastroenterology*, vol. 130, no. 4, p. A620.
- Burns, P. T. 1801, "Increasing Access to Work for Longer Term Community Mental Health Team clients: the impact of a work-placement training intervention Complete,".
- Collado, P., Richi, P., Revenga, M., Bachiller, F. J., Candelas, G., Blanco, M., Hernandez, C., Crespo, M., & Jover, J. A. 2001, "Low back pain and disability among working population: Efficacy of a specific return to work program", *Arthritis and Rheumatism*, vol. 44, no. 9, p. S188.
- Coste, J., Delecoeuillerie, G., Cohen de Lara, A., Le Parc, J. M., & Paolaggi, J. B. 1994, "Clinical course and prognostic factors in acute low back pain: an inception cohort study in primary care practice", *BMJ*, vol. 308, no. 6928, pp. 577-580.
- Deale, D. A. 1789, "CBT versus relaxation for chronic fatigue syndrome: outcome at 5 year follow-up Complete,".
- Fantom, M. A. 1789, "Chronic low back pain:- A controlled clinical trial to compare individual physiotherapy with attendance at a Back Rehabilitation Programme Complete,".
- Gignac, M. A. M., Cott, C., & Badley, E. M. 2002, "Adaptation to disability: Applying selective optimization with compensation to the behaviors of older adults with osteoarthritis", *Psychology and Aging*, vol. 17, no. 3, pp. 520-524.
- Indahl, A., Velund, L., & Reikeraas, O. 1995, "Good prognosis for low back pain when left untampered. A randomized clinical trial", *Spine*, vol. 20, no. 4, pp. 473-477.
- Kent, D. R. 1 A.D., "Vocational rehabilitation in stroke Complete,".
- Matas, A. J., Lawson, W., McHugh, L., Gillingham, K., Payne, W. D., Dunn, D. L., Gruessner, R. W., Sutherland, D. E., & Najarian, J. S. 1996, "Employment patterns after successful kidney transplantation", *Transplantation*, vol. 61, no. 5, pp. 729-733.
- Neau, J. P., Ingrand, P., Mouille-Brachet, C., Rosier, M. P., Couderq, C., Alvarez, A., & Gil, R. 1998, "Functional recovery and social outcome after cerebral infarction in young adults", *Cerebrovascular Diseases*, vol. 8, no. 5, pp. 296-302.
- Purdon S, Stratford N, Taylor R, Natarajan L, Bell S, Wittenburg D 2006, 'The Impacts of the Job Retention and Rehabilitation Pilot', DWP Research Report No. 342, DWP, London

Radford, D. K. 1 A.D., "Return to work following traumatic brain injury: Case Control study and economic analysis Ongoing,".

Straaton, K. V., Wrigley, J. M., White, M. B., Fine, P. R., & Maisiak, R. 1994, "Factors Associated with Return to Work (Rtw) among Persons with Work Disability Due to Arthritis and Related Musculoskeletal Disorders (Armd)", *Arthritis and Rheumatism*, vol. 37, no. 6, p. R30.

Study design (ie not an RCT or longitudinal method)

- Adam S, Emmerson C, Frayen C, Goodman A (2006), *Early quantitative evidence on the impact of the Pathways to Work pilots*, DWP Research Report No. 354
- Arthur S, Corden A, Green A, Lewis J, Loumidis J, Sainsbury R, Stafford B, Thornton P, Walker R (1999), New Deal for Disabled People: Early Implementation, DWP Research report No. 106
- Barnes H, Hudson M (2006), Pathways to Work: Qualitative research on the Condition Management Programme, DWP Research Report No. 346
- Blackburn V, Child C, Hills D (1999), New Deal for Disabled People: Early findings from the Innovative Schemes, DWP In-house Report No. 61
- Corden A, Nice K (2006), *Incapacity Benefit Reforms Pilot: Finding from the second cohort in a longitudinal panel of clients*, DWP Research Report No. 345
- Corden A, Nice K (2006), Pathways to Work: Findings from the final cohort in a qualitative longitudinal panel of incapacity benefits recipients, DWP Research Report No. 398
- Corden A, Nice K, Sainsbury R (2005), *Incapacity Benefit Reforms Pilot: Findings from a longitudinal panel of clients*, DWP Research Report No. 259
- Crowther, R. E., Marshall, M., Bond, G. R., Huxley, P. (2001). Helping people with severe mental illness to obtain work: systematic review. *British Medical Journal*, 322, 204-209.
- Dixon J, Mitchell M, Dickens S (2007), *Pathways to Work: Extension to existing customers* (*matched case study*), DWP Research Report No. 418
- Gardiner, J. (1997). Bridges from benefit to work: A review. Joseph Rowntree Foundation, York. *Downloaded from www.jrf.org.uk.*
- Kirby, S. & Riley, R. 2004, "Compulsory work-focused interviews for inactive benefit claimants: an evaluation of the British ONE pilots", *Labour Economics*, vol. 11, no. 4, pp. 415-429.
- Knight T, Dickens S, Mitchell M, Woodfield K (2005), *Incapacity Benefit reforms the Personal Advisor role and practices: Stage Two*, DWP Research Report No. 278

Layard, R., Clark, D., Knapp, M., & Mayraz, G. (2007). Cost-benefit analysis of psychological therapy. *National Institute Economic Review*, 202, 90-98.

Oxlade, L. 2006, "Fit for work", Physiotherapy Frontline, vol. 12, no. 2, pp. 14-17.

- Philo, C., Parr, H., & Burns, N. 2005, ""An oasis for us": 'in-between' spaces of training for people with mental health problems in the Scottish Highlands", *Geoforum*, vol. 36, no. 6, pp. 778-791.
- Waddell, G. & Burton, A. K. 2005, "Concepts of rehabilitation for the management of low back pain", *Best Practice & Research in Clinical Rheumatology*, vol. 19, no. 4, pp. 655-670

Outcome

- Chang D and Irving A, 2008 *Evaluation of the GP education Pilot: Health and Work in General Practice,* Department of Work and Pensions, Research Report 479, London
- Corden A and Nice K, 2006, *Pathways to Work: Findings from the final cohort in a qualitative longitudinal panel of incapacity benefit recipients,* Department of Work and Pensions, Research Report 398, London
- Dorsett, R. 2007, "The Effect of Pathways to work on labour", *National Institute Economic Review* no. 202, pp. 79-89.
- Feuerstein, M., Menz, L., Zastowny, T., & Barron, B. A. (1994). Chronic back pain and work disability: Vocational outcomes following multidisciplinary rehabilitation. *Journal of Occupational Rehabilitation*, 4 (4), 229-251.
- Franche, R. L., Cullen, K., Clarke, J., MacEachen, E., Frank, J., Sinclair, S., & Reardon, R. (2004). Workplace-based return-to-work interventions: A systematic review of the quantitative and qualitative literature. Institute for Work & Health. *Downloaded from www.iwh.on.ca.*
- Linton. S. J. (1995). Strategies for enhancing occupationally-oriented rehabilitation. *Journal of Occupational Rehabilitation, 5 (4), 203-206.*
- Staal, J. B., Rainville, J., Fritz, J., Van Mechelen, W., & Pransky, G. (2005). Physical exercise interventions to improve disability and return to work in low back pain: Insights and opportunities for improvement. *Journal of Occupational Rehabilitation*, 15 (4), 491-505.

Cost effectiveness: Excluded studies

- Abasolo L, Blanco M, Bachiller J, Candelas G, Collado P, Lajas C (2005), 'A health system program to reduce work disability related to musculoskeletal disorders', *Annals of Internal Medicine*, vol. 143, no 6, pp 404-414
- Abrams, D. (1993), 'The economics of return to work for survivors of traumatic brain injury: Vocational services are worth the investment', *Journal of Head Trauma Rehabilitation*, vol. 8, no. 4, pp. 59-76
- Ballegaard, S., Norrelund, S., & Smith, D. F. (1996), 'Cost-benefit of combined use of acupuncture, shiatsu and lifestyle adjustment for treatment of patients with severe angina pectoris', Acupuncture and Electro-Therapeutics Research International Journal, vol. 21, pp. 187-197
- Brewer M, Duncan A, Shephard A, Suarez M J, (2006), 'Did working families' tax credit work? The impact of in-work support on labour supply in Great Britain' *Labour Economics* 13(6):699-720.
- Campolieti, M. (2001), 'The Canada/Quebec Pension Plan Disability Program and the Labor Force Participation of Older Men', *Economics Letters*, vol. 70, no. 3, pp. 421-426.
- Canadian Coordinating Office for Health Technology Assessment (1992), 'Chiropractic treatment of neck and back disorders: a review of selected studies', *Canadian Coordinating Office for Health Technology Assessment*
- Cullberg, J., Mattsson, M., Levander, S., Holmqvist, R., Tomsmark, L., Elingfors, C., Wieselgren, I. M. (2006), 'Treatment costs and clinical outcome for first episode schizophrenia patients: a 3-year follow-up of the Swedish 'Parachute Project' and two comparison groups', *Acta Psychiatrica Scandinavica*, vol. 114, no. 4, pp. 274-281
- Fritz, J. M., Delitto, A., & Erhard, R. E. (2003), 'Comparison of classification-based physical therapy with therapy based on clinical practice guidelines for patients with acute low back pain: a randomised clinical trial', *Spine*, vol. 28, no. 13, pp. 1363-1371
- Gatchel, R. J., Polatin, P. B., Noe, C., Gardea, M., Pulliam, C., & Thompson, J. (2003), 'Treatment- and cost-effectiveness of early intervention for acute low-back pain patients: a one-year prospective study', *Journal of Occupational Rehabilitation*, vol. 13, no. 1, pp. 1-9
- Gill, C. R. W. (1996), 'Cost-effective rehabilitation', *Occupational Medicine*, vol. 46, no. 5, pp. 377-378

- Hunter, S. J., Shaha, S., Flint, D., & Tracy, D. M. (1998), 'Predicting return to work: a long-term follow-up study of railroad workers after low back injuries', *Spine*, vol. 23, pp. 2319-2328
- Jordan, K. D., Mayer, T. G., Gatchel, R. J. (1998), 'Should extended disability be an exclusion criterion for tertiary rehabilitation? Socioeconomic outcomes of early versus late functional restoration in compensation spinal disorders', *Spine*, vol. 23, no. 19, pp. 2110-2117
- Leon, A. C., Walkup, J. T., & Portera, L. (2002), 'Assessment and treatment of depression in disability claimants: a cost-benefit simulation study', *The Journal of Nervous and Mental Disease*, vol. 190, no. 1, pp. 3-9
- Miller, P., Kendrick, D., Bentley, E., & Fielding, K., (2002), 'Cost-effectiveness of lumbar spine radiography in primary care patients with low back pain', *Spine*, vol. 22, no. 20, pp. 2291-2297
- Monpere, C., Rajoelina, A., Vernochet, P., Mirguet, C., Thebaud, N. (2000), 'Return to work after cardiovascular rehabilitation in 128 coronary patients followed up for 7 years: results and medico-economic analysis', *Archives des Maladies du Coeur et des Vaisseaux*, vol. 93, no. 7, pp. 797-806
- Pinnington, M. A., Miller, J., Stanley, I. (2004), 'An evaluation of prompt access to physiotherapy in the management of low back pain in primary care', *Family Practice*, vol. 21, no. 4, pp. 372-380
- Seekamp, A., Regel, G., Tscherne, H. (1996), 'Rehabilitation and reintegration of multiply injured patients: an outcome study with special reference to multiple lower limb fractures', *Injury*, vol. 27, no. 2, pp. 133-138
- Shaw, S., Robertson, M.M., McLellan, R.K., Verma, S., & Pransky, G., (2006), 'A Controlled case study of supervisor training to optimise response to injury in the food processing industry', Work: A journal of prevention assessment and rehabilitation, vol. 26, no. 2, pp. 107-114
- Thomas, C. M. & Morris, S. (2003), 'Cost of depression among adults in England in 2000', *British Journal of Psychiatry*, vol. 183, pp. 514-519
- Turk, D. C. & Okifuji, A. (1998), 'Treatment of chronic pain patients: clinical outcomes, cost-effectiveness, and cost-benefits of multidisciplinary pain centers', *Critical Reviews in Physical and Rehabilitation Medicine*, vol. 10, no. 2, pp. 181-208
- Turk, D. (2001), 'Treatment of chronic pain clinical outcomes, cost effectiveness and cost benefits' *Drug Benefits Trends*, vol. 13, no. 9, pp. 36-38
- Turk, D. (2004), 'Chronic pain: purposes and costs of interdisciplinary pain rehabilitation programs, *Economics of Neuroscience*, vol. 4, no. 2, pp 64-69

Zeitzer, I. R. (1991), 'The Role of Assistive Technology in Promoting Return to Work for People with Disabilities: The U.S. and Swedish Systems', *Social Security Bulletin*, vol. 54, no. 7, pp. 24-29.

Appendix 5: Studies Pending

There were no effectiveness or cost effectiveness primary or review studies pending

Appendix 6: Example Completed Effectiveness Data Extraction and Quality Checklists

Data Extraction Form	Ref ID: Q4 Other 3569
Authors/Title/Source	!
M Desouza, M Sycamore, S Little, S G B Kirker Programme', Disability and Rehabilitation, Ap	'The Papworth Early Rehabilitation til 2007, 29 (8), 671-677
Project: LTSI	
Data extracted by: CM Dat	e of extraction: 19.11.07
Describe the study:	
Systematic review (including at least one RCT)	
Systematic review of experimental studies	
Systematic review of observational studies	
Randomised controlled trial: Individual	
Randomised controlled trial: Cluster	
Controlled non-randomised trial	
Controlled before-and-after	
Interrupted time series	
Before and after study	
Cross sectional (survey)	
Audit/Evaluation	

\checkmark

What was the research question?

What were the vocational outcomes of the Papworth Early Rehabilitation Programme which aims to get people on incapacity benefits into employment, fit for and seeking work, involved in voluntary work or education, and/or able to live independently?

Other study parameters:

Setting:

Geographical (City/country):

Cambridgeshire.

Social (school/workplace etc):

Papworth rehabilitation centres

Date of study (to/from):

Utilised Programme records from 1995 to 2003. Follow-up in November 2004. Who funded the study?

Not stated

Participants:

Number of participants:

Details on age, gender, and other characteristics (specifically disability, ethnicity, sexual orientation, religion or belief and socio-economic status) if presented:

87 male, 20 female

41 traumatic brain injury, 22 musculo-skeletal Injuries, 17 acquired brain injury, 15 back injuries, eight amputees and four other injuries

Were intervention groups balanced at baseline?:

Not applicable

Comments:

Individual

Sex: Not stated Age (range or mean): Not stated

Group: Not applicable

Organisation/Institution

Community/Environment

Policy/socio-political

Method of recruitment/enrolment and response rate:

All who started the programme between 1995 and December 2003

Method of allocation to intervention:

All participants allocated to intervention, no control group.

Selection criteria:

Inclusion:

All participants starting rehabilitation programme (107)

Exclusion:

Those who did not attend next recommended stage of acceptance process (132), those who were not recommended for acceptance (73), those who left programme before completion (13) and those who could not proceed as no

funding was available (20).

Intervention:

Description of the Intervention:

The Early Rehabilitation Programme run by the Papworth Trust, an 'employment, medical and life skills' rehabilitation programme for people with physical or cognitive disabilities. Delivered by team of a rehabilitation coordinator, case manager, job coach and assistant, vocational adviser, information technology assessor, and vocational psychologist. Help from a literacy tutor, occupational therapist and assistant, consultant in rehabilitation medicine, speech and language therapist and physiotherapist is bought in according to individual need.

Description of the comparator(s):

None

Method/mode of delivery (for example, peer education):

Job coaching, occupational therapy, medical rehabilitation.

Providers/deliverers of the intervention (including organisations involved):

The Papworth Trust

Length, duration and intensity of the intervention:

4-10 months, dependent upon the patient

Time to follow-up (average/median): Median 48 months

How many (n, %) participants completed the intervention?

94 of 107 (88%)

Details on age, gender, and other characteristics (specifically disability, ethnicity,

sexual orientation, religion or belief and socio-economic status) if presented:			
For non-completers, were the reasons for non-completers	tion described?		
No.			
Outcomes ¹ :			
Primary outcomes:			
Return to work (paid and unpaid) \checkmark			
Sustained return to work (paid and unpaid) \Box			
No impact on return to work \Box			
Were baseline measurements of outcomes assess	sed? Yes 🗸 No 🗆		
Describe outcome measures:			
Employment status			
Were baseline measurements of outcomes assess	sed? Not applicable		
Yes 🗆 No 🗆			
Were the outcome measure(s) validated?			
Yes 🗆 No 🗆 Not clear 🗆			
If yes, how?			
Secondary outcomes:			
Other work related outcome			
Acceptability of the intervention			
Adverse or unintended outcomes			
Barriers or facilitators of effective intervention			
Individual improvement in personal aspects	\checkmark		
Other - Describe:			
Work readiness			

¹ Adapted from Nutbeam's model (1998).

Describe outcome measures:				
Work ready' meant the individual had sufficient physical stamina, dexterity, basic cognitive skills (eg time-keeping) and inter-personal skills to be able to seek				
and sustain employment, with the help of schemes run by Disability				
Employment and Job Centres.				
Were baseline measurements of outcomes assessed? Not stated				
Yes 🗆 No 🗆				
Were the outcome measure(s) validated?				
Yes \Box No \Box Not clear \Box				
If yes, how?				
Analyses:				
Data collection methods used:				
Records and follow-up interviews				
Describe methods used (intention to treat, descriptive statistics, qualitative analysis				
etc): Not applicable				
Unit of analysis:				
Individual 🗸 Group 🗆 Organisation/institution 🗆				
$Community/cnvironment \square Policy/cocio political \square$				
Other (describe)				
Power				
Was a power calculation presented?Yes \Box No \checkmark				
If yes, describe:				

Was the study	y powered to detect	an effect if one exists?
Yes 🗆	No 🗸	Not clear 🗆
Any other pro	ocess details:	
Results:		
Briefly descri identified in t <i>example, absolu- confidence inte</i> population ch specifically di economic stat	be the results for eac he study? <i>List all me</i> <i>ute or relative risk, nu</i> <i>rvals that are provideu</i> naracteristics includi isability, ethnicity, so rus (if presented)?	th of the main outcomes (what size of effect is asures of effects in the units used in the study – for mber needed to treat, include p values and any d). Also describe results according to individual or ng age, gender, and other characteristics exual orientation, religion or belief and socio-
Of the 94 com	pleting the program	me at follow up there were:
56% (5 31% (3 4% (4) 2% (2) 2% (2)	 a) employed b) work ready in education, in voluntary work. living independent 	ly
Percer	itage and numbers o	f those with condition returning to work:
traum acquir muscu back in amput Other	atic brain injury 32% ed brain injury 41% llo-skeletal injuries 6 njuries 80% (12) ation 62% (5) 50% (2)	, (13) (7) 94% (14)

Are there any key criticisms of the conclusions drawn by the authors?

None

Does the paper address or offer any <u>evidence of effect</u> according to either of the following individual/population characteristics? If so, please ensure that evidence is presented in results above.

Older people	Yes □	No 🗸	Not clear \Box
Gender	Yes □	No 🗸	Not clear \Box
Ethnicity	Yes □	No 🗸	Not clear \Box
Socio-economic status	Yes 🗆	No 🗸	Not clear \Box
People with disabilities	Yes □	No 🗸	Not clear \Box

Other (please specify):

Does the paper demonstrate any evidence of harms or adverse effects associated with the intervention?

None

Do the authors identify any strengths and/or weaknesses of the evidence presented?

Weaknesses identified:

Incomplete entries in original records

Inconsistent formal follow up at 6 months

Loss of some clients from long-term follow up.

Rate of non attendance thought to be high but no comparable figures from elsewhere

No information about the employment outcomes of those who contacted the Trust but did not take up a place on the rehabilitation programme

In your opinion, are the results generalisable to the UK?

	Yes 🗸	No 🗆	Not clear 🗆
Why:			
]	Participants are c	lrawn from UK	wide referrals.
furthe	Do the authors ic er research?	lentify any evide	ence gaps or make any recommendations for
	Lack of a forma Patient drop ou	l outcome meast t needs to be ado	ure of handicap or participation dressed
Is the	re any data on co	st-effectiveness j	presented? Yes
	No formal cost four years incap Treasury rather period.	analysis but ave pacity benefit. In than beneficiary	rage cost of the programme is equivalent to dividual will become net contributory to y, so cost of programme saved in shorter
Are th	nere policy implic	cations of the wo	ork?
]	Funding for clien	ts was difficult t	to obtain from statutory services.
Are th	nere effective pra	ctice implication	s of the work?
	Yes, such initiativ and back to work	ves have the pote	ential to get employees off incapacity benefit

Methodology Checklist: Case studies (adapted from STROBE checklist, Version 3¹)

Study Identification	The Papworth Early Rehabilitation Pro	gramme:	
Include source (if	Vocational outcomes		
published-full	Disability and Rehabilitation, April 200	07, 29(8): 671-677	
reference details, if			
details of source).			
Cuidalina taniad TSI	Koy question no:04		
	Key question no. 04		
Checklist completed	CM		
by:	Civi		
INTRODUCTION			
Background/Rationale	Evaluation of programme of rehabilitation benefit recipients	n for incapacity	
Objectives	To evaluate the vocational outcomes of t	he programme	
METHODS		Bias present?	
Study design	Case studies	No comparison	
		group	
Setting	Eastern England, Papworth rehabilitation centres		
	Data originally gathered between 1995		
	and 2003, with a follow up in 2004		
Participants	Inclusion: All participants who started	No Comparison	
	the programme between 1995 and	group	
	2003.		
	Exclusion: Those who did not attend		
	next recommended stage of acceptance		
	process (132), those who were not		
	those who left programme before		
	completion (13) and those who could		
	not proceed as no funding was available		
	(20).		

¹ <u>http://www.strobe-statement.org/</u>

-

Variables of interest	Outcomos: omployment, work readiness	
	Outcomes. employment, work readiness	N
Measurement	Results measured as either	NO
	unemployed, returned to work, or	
	ready to return to work.	
	Although initial attendance was not for	
	the study, the 2004 follow-up was	
	carried out for the study. Study was	
	carried out by four researchers, two of	
	who work with the program being	
	evaluated.	
		NI / A
	If applicable, describe comparability of	N/A
	assessment methods across groups.	
Bias	Are identified sources of bias random or	N/A
	are they in one direction? Describe any	
	measures taken to address potential	
	sources of bias.	
Sample size	All participants who started the	
	programme between 1995 and 2003	
	(94).	
Statistical methods	(a) Describe all statistical methods	N/A
	including those to control for	
	confounding.	
	Describe how loss to follow-up and	They were not.
	missing data were addressed.	
		N
	Describe how any matching of cases and	Not addressed
	controls and missing data were	
	addressed.	
	If applicable, describe methods for	Divided by
	subgroup analyses and sensitivity	disability, length
	analyses.	of unemployment
Quantitative	Explain how quantitative variables are	Grouped according
variabies	analyzed eg. which groupings are	to disability and
	chosen, and why.	employment status
	Present results from continuous	N/A
	analyses as well as from grouped	
	anaryses, ir appropriate.	
Funding	Give source of funding and role of	Not known
Ĩ	funder(s) for the study and, if	
	applicable, the original study on which	
	the present article is based.	
RESULTS		
Participants	Report the numbers of individuals at	340 potentially

	each stage of the study (separately for cases and controls) (eg. numbers potentially eligible, examined for eligibility, confirmed eligible, included in the study, completing follow up, and analysed).	eligible, 107 started programme, 94 completed programme, and were followed up.
	Give reasons for non-participation at each stage.	None given.
Descriptive data	87 male, 20 female. 58 brain injuries, 22 Musculo-skeletal, 15 Back injuries.	Confounding bias present?
	Data on participant's employment status appears complete	
	Follow up for all participants occurred in 2004.	
Outcome data	53 employed, 2 in voluntary work, 33 ready for work and 4 were in education.	
OVERALL ASSESSMENT	 ++ Where all or most of the data is adequately described and where the conclusions of the study are thought very unlikely to alter (low risk of bias). + Where some of the data is adequately described and where the conclusions of the study are thought unlikely to alter (risk of bias). - Where few or no of the data is adequately described and where the conclusions of the study are thought unlikely to alter (risk of bias). 	-

Appendix 7: Economic Data Extraction and Quality Assessment Form

Q4 Data extraction and quality assessment for economic evaluations

Name of study Authors Journal details including year Country/countries Cross reference to data extraction for effectiveness evaluation (where applicable) Sample sizes of original studies (if applicable and if not cross referenced in 0.4 above)

1. Was a full economic evaluation undertaken? Did it include a comparative assessment of costs and health outcomes?

2. Describe the interventions, comparators, population, outcomes, perspective and time horizon included in the economic evaluation.

3. What form of economic evaluation was undertaken?Cost-effectiveness analysisCost-utility analysis

Cost-benefit analysis

Cost consequences analysis.

4. What type of modelling approach was used (cohort versus individual-patient level, dynamic versus static)? Is this appropriate to address the decision problem? What modelling methodology was used (within-trial evaluation, decision tree, markov, discrete event simulation, other)? Is this appropriate to address the decision problem?

5. Describe the key structural assumptions employed in the evaluation. Do these appear reasonable? What is the likely impact of these assumptions on the results of the evaluation?

6. Describe the assumptions surrounding the effectiveness data and the sources employed in the model. Were all relevant health outcomes included in the model? How were benefits measured and valued? What are the strengths and weaknesses of these data?

Quality of life measure used Quality of life for intervention Quality of life for comparator Number of QALYs gained for intervention Number of QALYs gained for comparator

7. Describe the assumptions surrounding the resource use and cost data employed in the model. Were all relevant costs included in the model? How were these measured and valued? What are the strengths and weaknesses of these data?

Year to which prices refer Total costs for intervention in original prices and UK 2007 prices Total costs for comparator in original prices and UK 2007 prices

8. Was discounting applied to costs and health outcomes to account for time preferences?

9. What were the results of the economic model? Were results presented incrementally? Are the base case results calculated using deterministic parameter values or the expected values?

10. Was a comprehensive uncertainty analysis undertaken? What methods were used to evaluate uncertainty (one-way, multi-way, probabilistic). How were the results of the uncertainty analysis presented (cost-effectiveness planes, cost-effectiveness acceptability curves)?

92 Interventions to Help Recipients of Incapacity Benefits Back to Work

10.1 Key results of the sensitivity analyses

11. Does the study report details of any model validation (concurrence of experts, internal/external consistency, predictive validity)?

12. What are the author conclusions? Does the study discuss the generaliseability of the results of the evaluation? Is it applicable to the UK setting?